Electronically Recorded

014766-18 30 Apr 2018 12:39:40 PM

SEAL

Book: 2018 Page: 014766-18

69 pages

REAL ESTATE DOCUMENT
GREENE COUNTY, MISSOURI
RECORDERS CERTIFICATION

Cheryl Dawson-Spaulding
Recorder of Deeds

Icunningham

First Amended
Declaration of Covenants, Restrictions, Easements,
Charges, Assessments, and Liens for

EAGLESGATE COMMUNITY ASSOCIATION

A master-planned community located in Greene County,
Missouri

Grantor: Eaglesgate, LLC, a Missouri entity

Updated
by the
Board of Directors
Eaglesgate Community
Association
And
Kevin Dunaway,
Neale & Newman, L.L.P.

© 2018 Eaglesgate Community Association, All Rights Reserved

Index

Section	<u>Title</u>	<u>Page</u>
ARTICLE I	DEFINITIONS	
ARTICLE II	ANNEXATION	
2.01	Annexation	
2.02	Right of Annexation	
2.03	Right of De-Annexation	10
ARTICLE III	EAGLESGATE COMMUNITY ASSOCIATION	
3.01	Powers and Duties of the Association	
3.02	Membership in the Association	
3.03	Board of Directors (Association Board)	
3.04	Suspension of Membership and Rights of Enjoyment	
3.05	Termination of Membership	17
3.06	Notice of Meetings and Referendums	. 17
3.07	Conduct of Association Elections.	
3.08	Limitation of Liability	
3.09	Area Associations	
3.10	Conflicts of Interest	
3.11	Voluntary Dissolution of the Association	20
ARTICLE IV	IMPOSITION OF ASSESSMENTS AND LIENS UPON PROPERTY	
4.01	Covenants for Assessments and Creation of Liens	20
4.02	Uniform Rate of Assessment.	
4.03	Rental Property	
4.04	Billing of Annual Assessments	
4.05	Commencement of Assessments	
4.06	Late Payments or Non-Payment	24
4.07	Enforcement Authority	25
4.08	User Fees and Charges	27
4.09	Additional Procedures	
4.10	Special Assessments	27
ARTICLE V	USE OF FUNDS	
5.01	Purposes for which Funds May be Used	28
5.02	Handling of Funds	29
5.03	Accumulation of Funds Permitted	29
5.04	Posting of Bond	29
5.05	Mortgaging of Common Property	29

ARTICLE VI	COMMON PROPERTY	
6.01 6.02	Use of Common Property	31
6.03 6.04	Maintenance of Common PropertySuspension of Rights	31 31
ARTICLE VII	DESIGNATION OF USE OF PROPERTY	
7.01	Change in Designation	32
ARTICLE VIII	DESIGN REVIEW COMMITTEE	
8.01	Purpose, Powers and Duties of DRC	32
8.02	Composition and Appointment	32
8.03	Officers, Subcommittees and Reimbursement	33
8.04	Operations of the DRC	33
8.05	Development Guidelines	35
8.06	Submission of Plans and Specifications	
8.07	Approval of Plans and Specifications	
8.08	Disapproval of Plans and Specifications	
8.09	Failure to Act	
		38
8.10	Inspection Rights	
8.11	Violations	
8.12 8.13	Fees	
ARTICLE IX	EASEMENTS	
9.01	Easements	39
9.02	Entry	
9.03	Conveying an Easement Area	
ARTICLE X	GENERAL RESTRICTIONS	
10.01	Maintenance Required by Owner	41
10.02	Land Use and Building Type	42
10.03	Landscape Restrictions	42
10.04	Building Locations	42
10.05	New Construction	42
10.06	Uncompleted Structures	42
10.07	Structures	43
10.08	Fences.	43
10.09	Placement of Signs on Property	43
10.10	Keeping of Animals	43
10.11	Disposition of Trash and Other Debris	44
10.11	Parking of Motor Vehicles, Boats and Trailers	45
10.12	Nuisances	47
10.13	Antennas Poles and Projections	47
1 1 / 1 	(2010-2017) A 1 000-2 700 1 1 000-2 10012	+ /

10.15	Subsurface Water	48
10.16	Drainage	48
10.17	Sanitary Sewers	48
10.18	Air and Water Pollution	48
10.19	Mining and Drilling	48
10.20	Placement of Pipelines	48
10.21	Commercial Activity	49
10.22	Fireworks and Use of Firearms	49
10.23	Laws and Ordinances.	49
10.24	"Off Road" Vehicular Traffic	49
10.25	Penalties for Violation of Article X	49
10.23	relatites for violation of ratiole re-	1,7
ARTICLE XI	RESIDENTIAL PROTECTIVE COVENANTS AND RESTRICTIONS	
11.01	Residential Provision.	49
11.02	Restrictions for Residential Lots.	49
11.02	Lot Use for Model Home or Real Estate Office	50
11.04	Use of Clothes Hanging Devices and Machinery	50
	Provisions Applicable to Lots Designated for Single Family	50
11.05		50
	Residences	30
ARTICLE XII	CONSTRUCTION ON LOTS	
12.01	Construction Standards	51
ARTICLE XIII	DURATION AND AMENDMENT	
13.01	Duration	53
13.02	Amendment.	53
13.02	Amendment	33
ARTICLE XIV	ENFORCEMENT	
14.01	Right of Action	54
14.02	Individual Assessments.	54
14.03	Specific Performance.	55
14.04	Enforcement of Liens.	55
		57
14.05	No Waiver	57
14.06	Additional Rules	57
14.07	Incorporation of Provisions in Deeds	37
ARTICLE XVI	MISCELLANEOUS	
15.01	No Reverter	58
15.02	Invalidity	58
15.03	Violation and Nuisance.	58
15.04	Violation of Law	58
15.05	Remedies Cumulative	58
15.06	Limitations	58
13.00	LillitatiOli5	20

15.07	No Personal Liability	58
15.08	View Lots	59
15.09	Assignability	59
15.10	Headings	60
15.11	Gender	60
15.12	Effect of Violation of Declaration on Mortgage	60
15.13	Delivery of Notices and Documents	60
15.14	Local Laws Not Superseded	60
15.15	Severability	60

EXHIBITS

Exhibit "A" LEGAL DESCRIPTION OF THE PROPERTY, Phase 1, 2, 3, and 4 COMMON PROPERTY

FIRST AMENDED

DECLARATION OF COVENANTS, RESTRICTIONS, EASEMENTS, CHARGES, ASSESSMENTS AND LIENS FOR EAGLESGATE

THIS ORIGINAL DECLARATION was made the 13th day of November, 2003, by Eaglesgate, LLC, a Missouri entity (hereinafter "Developer").

WHEREAS, this First Amended Delcaration is made this 30th day of April, 2018 by the Eaglesgate Community Association and

WHEREAS, the Developer created thereon a planned community to be known as Eaglesgate containing residential areas and related uses in substantial conformance with the governmental approvals therefore; and

WHEREAS, the Developer provided for the preservation of the values and amenities in said community and for the maintenance of common use areas, including any improvements located thereon: and, to this end, desires to subject the real property hereinafter described to the covenants, restrictions, easements, charges and liens hereinafter set forth; and

WHEREAS, the Developer deemed it desirable, for the efficient preservation of the values and amenities of said community, to create an agency to which should be delegated and assigned the power of maintaining, administrating and enforcing the covenants and restrictions and collecting and disbursing the assessments and charges hereinafter created; and

WHEREAS, the Developer incorporated under the laws of the State of Missouri, a not-for-profit corporation for the purpose of exercising the functions herein described; and

WHEREAS, the subdivision known as Eaglesgate has now matured and most of the lots now have homes constructed upon them, the Developer has passed away, and no person or business entity succeeded his business in developing the subdivision; and

WHEREAS, the Eaglesgate community has matured into a self-governing subdivision without the necessity of having a Developer take the initiative in managing any lots; and

WHEREAS, the Association believes it appropriate to amend the original Declaration of Covenants, Restrictions, Easements, Charges, Assessments and Liens for Eaglesgate into a First Amended Declaration; and

WHEREAS, the Association believes this Declaration be interpreted according to the plain, ordinary, and usual meaning of the words used here; and, further, that if there is any question about how this Declaration should be interpreted, the Declaration should be interpreted according to the purpose of the Declaration, which is to govern and regulate the existing character of the community.

NOW, THEREFORE, the Developer declared that the land described in <u>Exhibit "A</u> to be held, sold, used and conveyed subject to the following covenants, restrictions, easements, charges and liens, all of which are for the purpose of promoting the common good and general welfare of all Residents, Members and Owners and thereby enhancing and protecting the value, desirability and attractiveness of such land. These covenants, restrictions, easements, charges and liens shall run with such land and shall be binding on all parties having or acquiring any right, title or interest in such land or part thereof, subject to the limitations herein provided, and shall inure to the benefit of each Owner, his or its heirs, grantees, distributes, personal representatives, successors and assigns, the Association, each Resident, and the Developer.

<u>ARTICLE 1</u>

Definitions

The following terms, when used in this Declaration, or in any supplemental Declaration made effective against the Property according to law, and when the first letters thereof are capitalized, shall have the following meanings (except as otherwise expressly provided or unless the context otherwise requires):

- 1.01 <u>Annexation Property.</u> "Annexation Property" shall mean and refer to such real property owned or acquired by the Developer and not yet subject to this Declaration which is integrated by the Development Plan, and made subject to the scheme of this Declaration.
- 1.02 <u>Assessable Property.</u> "Assessable Property" shall mean and refer to the Property, together with all permanent structural improvements thereon, except such part or parts thereof as may from time to time constitute "Non-Assessable Property."
- 1.03 <u>Assessments.</u> The term "Assessments" shall have the meaning specified in Article IV, and shall include Annual Assessments and Special Assessments as such terms are defined in Article IV.
- 1.04 <u>Association.</u> "Association" shall mean and refer to the Eaglesgate Community Association, a not-for-profit Missouri corporation, or any successor thereof, charged with the duties and obligations set forth herein. "Area Association" shall have the meaning set forth in Section 3.09 hereof.
- 1.05 <u>Association Board.</u> "Association Board" shall mean and refer to the Board of Directors of the Association.
- 1.06 <u>Common Property.</u> "Common Property" shall mean and refer to the improved or unimproved real property, together with the Structures and personal property located thereon in which the Association owns interest as designated for the common use and enjoyment of the Owners and Residents, as such areas may be depicted on any recorded subdivision plat of the Property, or portion thereof, as "Private Open Space" and as shown on the attached <u>Exhibit "B</u>" or

as shown as common drainage easement, including but not limited to the landscaped portion of any street within the Subdivision, cul-de-sac islands, landscaped areas within any public street within the Subdivision, any private streets, entry roads, curb and gutter, sidewalks, gates and other improvements as shown on the recorded plat of the Property. Such interest or interests may include, without limitation, estates in fee, easements, leaseholds or licenses. Notwithstanding any legal presumption to the contrary, the fee title to any land shown upon any filed or recorded map or plat of any part of the Property which shall be designated as "Private Open Space" shall be until such time as the same shall be conveyed to the Association. The Common Property shall not contain any buildings used for commercial, residential or Living Unit purposes.

- 1.07 <u>Completed Unit</u>. "Completed Unit" shall mean and refer to a Living Unit upon which construction is completed and which has been or is, in fact, occupied.
- 1.08 <u>DRC</u>. "DRC" shall mean and refer to the Design Review Committee, which shall have the duties and functions specified in Article VIII hereof.
- 1.09 <u>Declaration</u>. "Declaration" shall mean and refer to this First Amended Declaration of Covenants, Restrictions, Easements, Charges and Liens, as the same may from time to time supplemented or amended in the manner prescribed herein.
- 1.10 <u>Deed</u>. "Deed" shall mean and refer to a deed, assignment or other recordable instrument conveying the fee simple title to a Lot or a recorded land sale contract, contract for deed similar instrument which requires the vendee to make periodic payments towards the purchase price for the purpose of eventually obtaining the fee simple title to a Lot.
- 1.11 <u>Development Guidelines</u>. "Development Guidelines" shall mean and refer to the rules, regulations and policy statements adopted, promulgated, revised and amended and enforced by the DRC pursuant to Article VIII of this Declaration.
 - 1.12 Director. "Director" shall mean and refer to a member of the Association Board.
- 1.13 <u>Easement Area</u>. "Easement Area" shall mean that real property or portion of real property described within an easement on the Plat, plats or maps filed or the be filed for record by the Developer in accordance with the Development Plan, and from time to time by recorded instrument, reserved for the easement purposes set forth in such instruments and as generally described in Article IX hereof and including access easement areas.
- 1.14 <u>Living Unit</u>. "Living Unit" shall mean and refer to any Structure or portion of a Structure situated upon any Lot designed and intended for use and occupancy as a residence by a single person, a family or a "family-sized" group of persons.
- 1.15 <u>Lot</u>. "Lot" shall mean and refer to any plot or parcel of land, including a condominium unit, shown on the Plat or plats or subdivision map of any part of the Property or any other lot or parcel of land, including a condominium unit, constituting part of the Property described in a Deed from the Developer or any subsequent Owner, which Deed has been recorded in the

Recorder of Deeds Office of Greene County, Missouri with the exception of Common Property, together with all permanent structural improvements thereon.

- 1.16 <u>Member</u>. "Member" shall mean and refer to every person or entity holding membership in the Association, as set forth in Article III.
- 1.17 <u>Non-Assessable Property</u>. "Non-assessable Property" shall mean and refer to all land designated "Private Open Space" or with a similar common property designation upon the Plat, any map or plats or any part of the Property.
- 1.18 <u>Non-Residential Property</u>. "Nonresidential Property" shall mean and refer to any Property or building or any portion of a building which has a nonresidential use and which is situated on Assessable Property. Structures constructed with Living Units, such as swimming pools, shall be considered "residential."
- 1.19 <u>Note</u>. "Note" shall mean and refer to all notes, bonds, debentures or other evidences of indebtedness issued and sold by the Association.
- 1.20 <u>Note Holder</u>. "Note Holder" shall mean and refer to the holder of any Note and all trustees and other representatives of any such holder.
- 1.21 Owner. "Owner" shall mean and refer to a person or entity holding record title to the fee interest of any Lot or Living Unit. "Owner" shall include a contract for deed seller, but shall exclude a person having an interest merely as security for the performance of an obligation.
- 1.22 <u>Plat.</u> "Plat" shall mean and refer to that certain Eaglesgate Plat recorded the 17th day November, 2003 in the office of Recorder of Deeds for Greene County, Missouri, together with all later phases recorded in respect of the Property.
- 1.23 <u>Private Open Space</u>. "Private Open Space" shall mean and refer to those areas of the Property designated as "Private Open Space" on any plat of the Property. The Developer has set aside the Private Open Space as depicted on the Plat.
- 1.24 <u>Property</u>. "Property" shall mean and refer to that certain real property described more particularly in <u>Exhibit "A"</u> attached hereto and made a part hereof, together with such Annexation by annexation pursuant to Article II hereof.
 - 1.25 <u>Resident</u>. "Resident" shall mean and refer to any person who:
 - (a) owns a Living Unit within the Property and has manifested his present intent to reside in that dwelling even though he may be temporarily absent; or
 - (b) is actually living within the Property in the same household with a person described in either Section 1.28(a) or in Section 1.28(c), whether or not he is a member of the immediate family of such person; or

- (c) is a Residential Tenant.
- 1.26 <u>Residential Area.</u> "Residential Area" shall mean and refer to Lots 1 through, 210 Eaglesgate Plat, together with Lots within the Annexation Property which may be specifically designated as "Residential Area" in any Declaration of Annexation.
- 1.27 <u>Residential Tenant</u>. "Residential Tenant" shall mean any person who occupies a Living Unit as the named "lessee" under a written lease from an Owner and which lease has been filed with the Association and which lessee has been registered with the Association as herein provided and required.
- 1.28 <u>Restriction</u>. "Restriction" shall mean and refer to any covenant, restriction, easement, lien or other obligation created or imposed by this Declaration.
- 1.29 <u>Right of Action</u>. "Right of Action" shall have the meaning specified in Article XV hereof.
 - 1.30 Structure. "Structure" shall mean and refer to:
 - (a) any thing or object, trees and landscaping, the placement, size, shape, color, height and quality of which upon any Lot may affect, in the opinion of the DRC, the appearance of such Lot, including by way of illustration and not limitation, any building or part thereof, garage, porch, greenhouse or bathhouse, covered or uncovered patio, swimming pool, fence, curbing, paving, wall, fence, hedge, sign, appurtenance, or any temporary or permanent improvement to such Lot; and
 - (b) any excavation, fill, ditch, diversion dam, retention basin or other thing or device which affects or alters the flow of any waters in any natural or artificial stream, wash or drainage channel from, upon or across any Lot; and
 - (c) any change in the grade of any Lot of more than six (6) inches.

ARTICLE II

Annexation

- 2.01 <u>Annexation</u>. All of the property owned or acquired by the Developer has been conveyed either to the Association or to Owners. Those properties are described in Exhibits "A" and "B". The only manner in which any additional land can be subjected to this Declaration shall be by and in accordance with the procedure set forth in Section 2.02 or in Section 2.03.
- 2.02 <u>Right of Annexation</u>. The Association may annex additional land to the Property by recording a Declaration of Annexation. Any such annexation shall require the approval of Members voting in person, or by proxy at an Association meeting at which a quorum is present, or voting in a

referendum on such issue after proper notice is given, each in accordance with Section 3.07. The Association may subject such annexation property to another set of covenants, restrictions, easements, charges, and liens.

- 2.03 <u>Right of De-Annexation</u>. The Association upon an election conducted under provisions of Section 3.07, may de-annex and release any portion of the Property or any portion of the Annexation Property from this Declaration, from any Declaration of Annexation and from the jurisdiction of the Association, provided that:
 - (a) there has been first filed with the Association Board a written request for deannexation (the "De-annexation Request") which shall describe by legal description and by street address, if available, all of the Property with respect to which de-annexation is requested;
 - (b) notification of the receipt of the De-annexation Request has been given according to the notice provisions of this Declaration to all Owners at least sixty (60) days prior to any vote by the Members with respect thereto;
 - (c) each Owner of a Lot or Living Unit, and including any lienholder on any such Lot or Living Unit whose consent may need to be obtained, situated within the boundaries of the Property proposed to be de-annexed and released has executed a written statement of consent to such action, which statement of consent shall be in such form as deemed appropriate by the Association Board.

Upon a vote under the provisions of Section 3.07, to de-annex and release such Property from this Declaration, from any Declaration of Annexation and from the jurisdiction of the Association, the Association Board shall cause to be filed in the appropriate Office of the Recorder of Deeds, a written release describing the Property to be released. Such release document shall be signed by all of the Owners of the Lots and Living Units within the boundaries of the Property to be de-annexed and shall contain such covenants, terms and conditions as the Association Board deems appropriate.

ARTICLE III

Eaglesgate Community Association

3.01 Powers and Duties of the Association. The Association is organized to operate for the common good and general welfare of the Residents, Members and Owners and consistent therewith, to acquire, own, improve, maintain, preserve, convey and control the Common Property, to administer and to enforce all covenants, restrictions, easements and charges contained in the Declaration and all liens created herein, and otherwise to promote the health, safety and general welfare of the people of said community, and to do and perform any and all acts which may be necessary or proper for or incidental to the exercise of any of the express powers (but not intended as affirmation obligations unless so stated) of the Association, including by way of illustration and not obligation, unless so stated, or limitation:

- (a) <u>Assessments</u>. The Association may levy Assessments on the Owners and enforce payment of such Assessments, all in accordance with the provisions of the Declaration set forth in Article IV.
- (b) Right of Enforcement. The Association shall also have the power and authority from time to time in its own name on its own behalf or on behalf of any Owner or Owners who consent thereto, to commence and maintain actions and suits to restrain and enjoin any breach or threatened breach of the Declaration and to enforce, by mandatory or prohibitive injunction or other legal and equitable relief, all of the provisions hereof to pursue its Right of Action as provided in Article XV herein. The Board of Directors is vested with the decision of whether to pursue a right of enforcement, and how it should be pursued, tried, settled, or otherwise resolved under its authority under Section 3.03.
- (c) Rules and Enforcement. The Association, through the Board of Directors, may adopt, promulgate, amend, revoke, and enforce rules and regulations regarding all of Eaglesgate's facilities, features, common grounds, events and programs, including the clubhouse, pool, tennis and basketball courts, ponds, and all other such features as may be developed and offered for the use of Eaglesgate Members and guests. If a resident violates such rules and regulations, the Association shall have the power and authority to pursue its Right of Action as provided in Article XV hereof, together with all remedies, whether at law or in equity, and including but not limited to the remedy of injunctive relief and obtaining a monetary judgment for all costs, expenses, including reasonable attorney's fees, and damages.
- Common Property. The Association may plan and implement community (d) programs and conduct Association programs on or in Common Property. Further, the Association may plan, design, acquire, improve, construct on, lease and equip the Common Property with, by way of example and not limitation or affirmative obligation, parks and other open space, trees, flowers, other landscaping, fountains, benches, shelters, public sculpture, pedestrian pathways, ornamental walls, lighting systems for such pathways, bridges or underpasses for such pathways, retention basins, tot lots, playgrounds, pools and other recreational facilities, office space, storage and maintenance buildings, garages and other buildings and facilities deemed necessary or desirable by the Association Board (collectively, the "Common Property Improvements" which may be referred to herein with the Common Property as the Common Property). The Association may also enter into contracts, leases or rental agreements for the purpose of providing such recreational facilities as deemed necessary or desirable by the Association Board and shall maintain, repair and replace the Common Property Improvements and provide adequate comprehensive insurance for the Common Property and Common Property Improvements, all as shall be

determined to be necessary by the Association Board. The Association may convey and dedicate to the City or County and to the use and enjoyment of the general public any of the Common Property, subject to the approval of, and the appropriate acceptance by, the City or County.

- (e) Easements and Rights-of-Way. The Association may grant and convey easements and rights-of-way in, on, over or under the Common Property and the Property for the purposes of constructing, erecting, operating or maintaining thereon, therein or thereunder (i)overhead or underground lines, cables, wires, conduits or other devices for the transmission of electricity and for lighting, heating, power, telephone, community television, radio and audio antenna facilities and other purposes, (ii) public sewers, storm water drains and pipes, water systems, sprinkling systems, water, heating and gas lines or pipes, and (iii) any similar public or quasi-public improvements or facilities as may be considered necessary for the common good of said community.
- (f) Employment of Agents. The Association may employ the services of any person or corporation as manager (herein, "Manager"), together with other employees, to, as may be directed and delegated by the Association Board, manage, conduct and perform the business, obligations and duties of the Association and may enter into contracts for such purpose.
- (g) <u>Insurance</u>. The Association shall obtain and keep in force such policies of insurance and surety bonds, as are necessary to adequately insure and protect the Common Property and the operations thereon and of the Association and as deemed by the Association Board to be necessary and appropriate.
- (h) <u>Management of Improvements</u>. The Association shall manage and control for its Members all improvements within public right of ways and on the Common Property, provided that such management and control of said improvements shall at all times be subject to that had and exercised, if any, by the City, the appropriate county and the State of Missouri.
- Landscape Maintenance. Owners and prospective owners of Lots, and the public in general, are hereby placed upon notice of the regulations set forth by the County of Greene, State of Missouri concerning maintenance of landscaping and that Greene County requires the following provision within this First Amended Declaration and within this subsection. If the regulations of Greene County concerning landscape maintenance are amended after the effective date of the First Amended Declaration, this First Amended Declaration shall be interpreted to confirm to that regulation. The Association shall, subject to the approval of the City or County, care for, spray, trim, protect, provide irrigation for, and replant shrubbery, ground cover and trees on all streets, and on islands located therein, on the Common Property at the Subdivision entrance, medians, or other landscaped areas within any right-of-

way of any public street located within the Subdivision and any Lot, if necessary, and shall care for, irrigate, protect and replant shrubbery, re-sow grass and replace sod in the Private Open Spaces and Common Property, where the maintenance thereof is for the welfare and benefit of the Residents, Members and Owners in the judgment of the Association Board and to the extent that the Greene County Highway Department deems necessary to maintain public safety. The Board of the Association shall be the sole judge as to the appropriate maintenance of all grounds within any Common Property, except any landscaped or planted areas within the right-of-way of a public street or any common area drainage easement depicted on the final plat shall be maintained to the satisfaction of the Greene County Highway Department and the Greene County Resource Management Department, respectively. In the event landscaping within any right-of-way or common area drainage easement shall not be maintained by the Association to the satisfaction of the Greene County Highway Department or Resource Management Department, the County shall provide the homeowners association, or its last known registered agent, with written notification of any deficiencies, whereupon, the Association shall have thirty (30) days to correct any deficiencies. In the event the Association fails to correct any deficiencies in landscaping of road right-of-ways or common area drainage easements within the Subdivision as delineated by Greene County officials within thirty (30) days of receipt of notice, then in that event the County may either (1) have the maintenance performed and the homeowners association shall be billed for the cost of said maintenance, or (2) the county my remove the Landscaping, median or landscaped area within any right-of-way in said subdivision. Except as otherwise specifically provided, any expense of the Association for administration, maintenance, operation, repair or replacement of any common area drainage easement and landscaping within any public right-of-way shall be treated as and paid for as a common expense of the Association, provided, however, if the County is not reimbursed for the costs of maintaining common area drainage, easements or landscaping within any right-of-way on the Property by the Association, Greene County may assess the cost of such maintenance or abatement in the same manner as assessments are levied by the Association, and the same shall be a levy and a personal liability to each lot owner in the Subdivision to the same extent as other assessments provided in Article IV. In the event the property is annexed into the City of Springfield, Missouri, the City of Springfield shall succeed to the rights of Greene County hereunder. All areas shown as Common Property on the final plat of the Property and all common area drainage easements shown thereon shall be owned, kept and maintained by the Developer until such time as the Developer shall determine to convey the same to the Association which thereupon shall accept delivery of said conveyance and shall thereafter hold all such areas as Common Properties and be responsible for such maintenance.

- (j) <u>Maintenance of Vacant Property</u>. The Association may mow, care for, maintain and remove rubbish from vacant or unimproved Property (except those Lots on which construction has commenced), and do any other things necessary or desirable in the judgment of the Association Board to keep any vacant and unimproved Property and the parking in front of any Property neat in appearance and in good order.
- (k) <u>Street Lighting</u>. The Association shall provide such lights as the Association may deem advisable on streets and sidewalks, in Private Open Space, gateways, entrances, or other features, and on other Common Property or public property subject to the prior written approval of the DRC.
- (l) <u>Snow Removal and Street Cleaning</u>. The Association may provide for the removal of snow from sidewalks and streets and cleaning of streets, gutters, catch basins, sidewalks and pedestrian ways, and for repair and maintenance of sewers, storm sewers and appurtenant drainage facilities.
- (m) <u>Signs</u>. The Association may erect and maintain signs, after such signs are approved by appropriate public authorities and after such signs are approved in writing by the DRC.
- (n) <u>Security Protection</u>. The Association may employ duly qualified officers or implement electronic security systems for the purpose of providing such security protection as the Association Board may deem necessary or desirable in addition to the protection rendered by public authorities.
- (o) Acquisition of Real Estate. The Association shall acquire and own title to such real estate as may be reasonably necessary to carry out the purpose of the Association and promote the health, safety, welfare and recreation of Owners; pay taxes on real estate and facilities owned by it; and pay such taxes as may be assessed against the Common Property.
- (p) <u>Retention Ponds</u>. The Association shall maintain the three ponds within the community. The Association Board shall formulate and publish rules regarding use of the ponds and enforce those rules.
- (q) <u>Bylaws.</u> The Association shall have Bylaws to guide and govern the Board of Directors in performing its duties and responsibilities as listed herein The Board of Directors shall develop the Bylaws and amend as needed. The purpose of the Bylaws shall be to insure an effective and efficient transition from one Board to the next and a consistent and equitable administration of the Association's continuing operations.

Consistent with the foregoing, the Association is authorized to exercise all powers which a corporation organized under the Not-For-Profit Corporation Law of Missouri may exercise.

3.02 <u>Membership in the Association.</u>

- (a) Each Homeowner residing in Eaglesgate shall be entitled to one (1) Membership in the Association, notwithstanding the number of Lots owned. Such owner shall be entitled to one vote for each Lot owned, so long as that owner remains an Owner of such Lot(s), pays an assessment on each Lot, and is in good standing.
- (b) Each non-resident Owner of property in Eaglesgate, notwithstanding the number of Lots owned, shall be entitled to one (1) Association Membership and one (1) vote in the Association, so long as such Owner specifies in writing to the Association the name of the individual who holds the Association Membership. In the absence of such written specification, Assessments shall be charged against the Lot and Owner thereof, but there shall be no right to vote the Membership. The Member must be an individual who is either an Owner, or if the Owner is or includes another individual, the Member may be an individual who is a partner if the Owner is or includes a partnership, or an officer of a corporation if the Owner is or includes a corporation, or a beneficiary of a trust if the Owner is or includes a trust, or an owner of an entity if the Owner is or includes a person other than an individual, a partnership, a corporation, or a trust. Anything in this subsection to the contrary notwithstanding, where a Lot is owned of record in any manner of joint or common ownership, the joint or common Owners thereof shall share among them the rights (including voting rights) given to an Owner pursuant to this Declaration, which they shall be entitled to exercise as a whole, but not in part, in whatever manner they shall jointly determine. With respect to voting rights in particular, joint or common ownership of a Lot shall entitle the Owners thereof to a total of one (1) vote, to be exercised in whatever manner they shall jointly determine. If the joint or common Owners cannot agree upon a vote, then no vote shall be counted and any attempt to cast a vote given that disagreement shall be considered void.
- (c) A non-resident Homeowner may designate in writing to the Association the name of a person residing in the non-resident Owner's home (lessee, renter, tenant) who shall be entitled to one (1) vote in Association elections.
- (d) A builder of a residence on a Lot, although an Owner, shall not be entitled to any vote in the Association unless and until such builder occupies the Living Unit as such builder's place of residence.
- (e) Subject to the provisions of this Section 3.02, once a Member has been specified as an Owner, a successor Member may only be specified as such Owner upon at least fifteen days' prior notice to the President of the Association; provided, however, the foregoing shall not impair the provisions of Section 3.02(f).

- (f) A Membership shall not be transferred, pledged or alienated in any way, except as herein expressly provided. Subject to the provisions of Section 3.02(a). an Association Membership shall automatically be transferred to a new Owner upon the transfer of the Lot to which it appertains (and then only to such transferee), whether by sale, intestate succession, testamentary disposition, foreclosure of a mortgage or other legal process transferring fee simple title to such Lot.
- (g) Subject to the provisions of this Declaration and the Association's By-Laws, the Association Board may make, amend or rescind such rules and regulations as it deems advisable for any meeting of Members, Association vote, referendum or election.

3.03 Board of Directors (Association Board).

- (a) The powers of the Association shall be vested in, exercised by, and under the authority of, all the affairs of the Association in accordance with the Association Articles of Incorporation and By-Laws, and shall be controlled by a Board of Directors consisting of five (5) persons who shall be Members (the "Association Board"). The Association Board, by a majority vote, shall exercise for and on behalf of the Association all powers, duties and authority vested in or delegated to the Association.
- (b) Directors, except for Directors appointed or elected pursuant to Section 3.09 hereof, shall be elected so that two (2) Directors and three (3) Directors shall be elected respectively in alternating years. Directors shall be elected for two-year terms of office and shall serve until qualified successors are elected or appointed.
- (c) Directors shall not discriminate against any Member, resident, tenant, or user of Common Property on the basis of race, color, sex, religion, national origin, family composition, or marital status. Further, the Board, in the exercise of its power granted pursuant to this Declaration, shall not take any action which is intended to or does, in effect, discriminate against persons of a particular race, color, sex, religion, national origin, family composition, or marital status.
- (d) The Board of Directors shall also take action upon recommendations of the Design Review Committee after that committee has a full opportunity to consider issues within its authority. It is hereby acknowledged the Directors shall have full authority to so proceed on behalf of the Association once the Design Review Committee makes its recommendations.
- 3.04 <u>Suspension of Membership and Rights of Enjoyment</u>. The Association Board may Suspend the voting rights of Members and the rights of enjoyment, except as to the Private Open

Space and to the lakes of any Member, Resident or user of the Common Property and the services offered thereon who:

- (a) is subject to a Right of Action by reason of having failed to take reasonable steps to remedy a violation or breach of the Declaration within the number of days specified in a written notice given by the Association Board after such violation or breach; or
- (b) has allowed any Assessment levied by the Association pursuant to this Declaration to become delinquent; or
- (c) has failed to pay any user fee or charge levied by the Association when due and payable; or
- (d) has violated any rules and regulations adopted by the Association Board governing the use and enjoyment of the Common Property or services thereon.

Such suspension shall be for the balance of the period in which the conditions set forth in subsections (a), (b), (c) and (d) of this Section 3.04 exist.

- 3.05 <u>Termination of Membership</u>. No owner shall continue to be a Member after he ceases to hold a qualifying interest in any Lot or Living Unit. No Member may avoid his obligations under this Declaration by declining to use Common Property, abandoning his Lot, or by any other act of abandonment or renunciation.
- 3.06 <u>Notice of Meetings and Referendums</u>. Proper notice shall be given by the Association Board of all meetings of the Association Board at least fifteen (15) days prior to the meeting date; and of all meetings of the Association Members, public hearings or referendums at least thirty (30) days prior to the hearing or referendum. The methods and procedures of such notice shall be determined by the Association Board in accordance with the By-Laws of the Association.
- 3.07 <u>Conduct of Association Elections</u>. Any other language herein notwithstanding, the following provisions shall govern the conduct of elections in the Eaglesgate Community Association. A community vote on such matters as changes to the regular and special assessments, amendments to this Declaration of Covenants, election of Directors to the Association Board, and other significant issues as ordered by the Association Board, shall be considered valid only if they are conducted in accordance with the following provisions.
 - (a) Eaglesgate residents shall be notified of all upcoming elections at least two notices before the election. Where required elsewhere in this Declaration, a community meeting will be scheduled by the Association Board to provide an opportunity for open discussion of the subject of the election. Notification of elections and community meetings may be made by mail, email, Web site posts, or a combination of these or other means, as determined by the Board of Directors.

- (b) All Eaglesgate Community Association elections shall be by written ballot distributed by mail, with stamped and addressed return envelope, to each member of the Association.
- (c) To be an eligible voter in any and all Eaglesgate Community Association elections, a resident must be a homeowner residing in Eaglesgate and a member of the Association in good standing. Members are not eligible to participate in community elections if they have had their membership or common property privileges suspended or terminated, either for violations of this Declaration; failure to pay assessments, liens, or penalties; or for other reasons determined by the Association Board. Builders and non-resident homeowners are not eligible to vote in Association elections. Non-resident homeowners involved in joint or common ownership are entitled to one vote if a person has been designated in writing, as provided in Section 3.02. Tenants may be eligible to vote if they are so designated in writing as provided in Section 3.02
- (d) An eligible voter may cast only one vote, except that a resident homeowner who owns and pays assessments on multiple lots may cast one vote for each of those lots.
- (e) An election shall be deemed valid if at least one-half-plus-one of the eligible voters cast written ballots. An issue shall be decided if at least two-thirds of the ballots cast are in favor. An election for members of the Board of Directors shall be determined by the two or three candidates receiving the highest number of votes cast by members voting in the election.
- (f) Votes shall be counted by staff of the Association's management company and monitored by two Members selected from the community, or by two Directors of other Associations, or by others selected by the Eaglesgate Board to assure a fair and credible count. The vote shall be kept confidential until communicated to the Board of Directors, which then will inform the community. All ballots will be kept confidential and retained by the management company should questions should arise.
- (g) Voting by proxy or in person at a community meeting is permitted only if so ordered by the Board of Directors for specific, unique, and/or exigent circumstances. A meeting quorum must consist of a minimum of one-half-plus-one of the Association's eligible voters.
- 3.08 <u>Limitation of Liability</u>. No member of the Association Board, officer of the Association, member of any committee of the Association, whether such committee is specifically described in this Declaration or hereafter created by the Association, shall be personally liable to any Owner, Member or Residential Tenant or to any other party, including the Association, for any damage, loss or prejudice suffered or claimed on account of any act, omission, error or negligence of such Association Board, officer to the Association, committee member, or of the Manager, if

any, any other representative or employee of the Association or of Developer, provided that such person has, upon the basis of such information as may be possessed by him, acted in good faith, without willful or intentional misconduct.

- 3.09 <u>Area Associations</u>. Certain areas of the Property may encompass common facilities and designated uses, the governance of which by the Association, may be benefited from informal organizations whose members would be Owners of Lots and Living Units encompassed by such areas. As may be determined by the Developer, and in order to aid the Association in fulfilling its duties hereunder, the Developer may, in its sole discretion and in order to so assist the Association, establish a localized and informal association of such Owners, which informal association may be shown on any subdivision plat of the Property as an "Area Association."
 - (a) <u>Membership</u>. Any Member of the Association who owns a Lot or Living Unit within an Area Association shown on any subdivision plat shall, be virtue of such ownership, also be a member of the Area Association created for such area.
 - (b) <u>Purpose</u>. The Area Association will be an informal organization of Owners who may from time to time as such Owners deem appropriate convene informal meetings in order to discuss Association business and the interests of the Owners in and to the Lots and Living Units located within the Area Association. The Area Association will operate to promote the common good and general welfare of the Owners in the Area Association consistent with this Declaration and the Articles, By-Laws and rules and regulations of the Association.
 - (c) <u>Informal Organization</u>. No Area Association will be incorporated nor in any other way formally organized but nevertheless may conduct meetings and otherwise pursue the common objectives of the owners in the Area Association consistent with this Declaration and the Articles, By-Laws and rules and regulations of the Association.
 - (d) <u>Superior Jurisdiction of the Association</u>. The Association shall have jurisdiction over all Area Associations and over all of the Property and every Owner who shall be a Member of the Association. Membership in an Area Association shall not grant any greater or lesser right to any Owner or Member, as a result of such membership, than such Owner or Member has as an Owner of any Lot or Living Unit on the Property.
- 3.10 <u>Conflicts of Interest</u>. An elected or appointed member of any Association board or committee, or similar entity, representing or acting on behalf of Eaglesgate residents may not participate in any decision on a matter in which that person has a direct or indirect financial interest, or on a matter in which that person has personally provided professional consulting services for a fee to any party that has submitted an application or request to an Association entity.

- (a) If two or more members of an Association board or committee are disqualified from participating is a request or application because of a conflict of interest, the Association's Board of Directors shall name substitute members to act only on the matter resulting in the disqualification. For a period of one (1) year after service on such an entity, no former member may represent any matter before that entity when that member has participated in any decision on said matter.
- (b) Each member of an Association board or committee shall inform the appropriate body in writing of any direct or indirect financial relationship with any applicants, builders or developers with an ownership interest in any Lot or Living Unit, or vendors providing a service or product to Eaglesgate. Such disclosure shall be made within fifteen (15) days of knowledge of the establishment of the relationship or commencement of a member's service on the board or committee, and shall be available for inspection by the Association Board and all Members.
- (c) Each member of an Association board or committee shall execute a conflict of interest letter at the time the member begins service and each year thereafter throughout the time of service. The letter statements to be signed will be provided by the Association. Executed statements will remain on file for a minimum of two years after service ends.
- 3.11 <u>Voluntary Dissolution of the Association</u>. The Association may not be voluntarily dissolved without the prior written consent of Greene County, Missouri or the City of Springfield, Missouri in the event the Property has been annexed into the City of Springfield, notwithstanding any contrary provisions in its Articles of Incorporation or By-Laws.

ARTICLE IV

Imposition of Assessments and Liens Upon Property

- 4.01 <u>Covenants for Assessments and Creation of Liens</u>. The Association and each Owner, jointly and severally, for himself, his heirs, distributes, legal representatives, successors and assigns, by acceptance of a Deed or other conveyance for any Lot which is Assessable Property, whether or not the covenants contained herein shall be expressed in any such Deed or other conveyance, hereby covenants and agrees that:
 - (a) he will pay to the Association all Assessments which may or shall be levied by the Association against Assessable Property owned by him in each year or any part thereof, and that he will pay to the Association the user fees and charges and all other duly authorized charges to be established as herein provided, if applicable, levied by the Association in each year and including Special Assessments levied pursuant to Section 4.09 hereof;

- (b) he shall be personally liable for all such Assessments and user fees and charges which become due while he is the Owner of each Lot or Living Unit being assessed;
- (c) all Assessments, together with the continuing obligation to pay each Assessment assessed in all future years, and all user fees and charges, together with all costs expenses, interest and reasonable attorney fees incurred in the collection of delinquencies, shall become, upon the filing of this Declaration, and thereafter remain a charge against and be secured by a continuing lien upon the Assessable Property of such Owner; and
- (d) said charge and lien shall be superior to any and all other charges, liens or encumbrances which may hereafter in any manner arise or be imposed upon the Assessable Property (or the Non-assessable Property to the extent that it may later become Assessable Property) whether arising from or imposed by judgment or decree or by any agreement, contract, mortgage or other instruments, excepting only:
 - (i) purchase money mortgages or deeds of trust given to finance the purchase of the Lots subject to the mortgage or deed of trust or to finance construction of improvements on the Lot subject to the mortgage or deed of trust; provided, however, that this subordination to such mortgages shall apply only to Assessments which have become due and payable prior to a sale or transfer of any Lot on account of the foreclosure of any such mortgage or on account of any other proceeding in lieu of foreclosure; such a sale or transfer at foreclosure or in lieu of foreclosure shall not release such Lot from the lien or relieve the new Owner, his successors and assigns from liability for any Assessments thereafter becoming due; and
 - (ii) such liens for taxes or other public charges as are made superior by applicable law; provided, however, that this subordination to such liens shall apply only to Assessments which have become due and payable prior to a sale or transfer of any Lot on account of the foreclosure of any such lien or on account of any other proceeding in lieu of foreclosure; such sale or transfer at foreclosure or in lieu of foreclosure shall not release such Lot from the lien of or relieve the new Owner, his successors and assigns from any liability for any Assessments thereafter becoming due.

Nothing contained in this subsection (d) of Section 4.01 shall be construed so as to constrain or impair the payment of funds realized from a foreclosure sale, to the extent of any funds remaining after satisfaction of prior liens; in order of priority to holders of subordinated liens.

4.02 <u>Uniform Rate of Assessment.</u>

- (a) For the purpose of providing funds for the uses specified in Article V hereof, the Association Board shall assess against the Assessable Property in each year, beginning with the year in which Common Property is first transferred to the Association, a charge (referred to herein as "Assessment" or "Annual Assessment"), which shall be uniform with respect to all Assessable Property within each classification of Assessable Property, as hereinafter provided, and shall be in such amounts as determined by the Association Board. The Association may divide all Assessable Property into classifications, which classifications shall be based upon the character of ownership nature of use, i.e., residential type, status of occupancy and such other criteria as the Association Board may deem pertinent. Lots or Living Units which are not occupied by a Resident and which are owned by a builder shall be assessed at a rate to be determined by the Association Board.
- Not later than ninety (90) days prior to the end of each fiscal year for the (b) Association, the Association Board shall prepare an annual cash budget projecting anticipated revenues, cash receipts, cash expenditures, and net cash, surplus or deficit for the ensuing fiscal year (the "Association Budget"). The fiscal year for the Association shall be the calendar year. The proposed Association Budget will automatically be approved unless two-thirds (2/3) of the Members vote not to approve such Association Budget under provisions of Section 3.07. Upon approval of the Association Budget, the Association Board will determine the manner in which Assessments are to be made; provided, however, that the Assessments will be made on a per Living Unit basis and not on market value or assessed value. The Association Board may not increase Annual Assessments to any Living Unit by more than fifteen percent (15%) of the previous year's Assessment for such Living Unit unless the Association Board determines that a greater adjustment is necessary solely as a result of inflation. The rate of Assessment for an individual Lot can change as the character of ownership, nature of use and the status of occupancy of said Lot changes, which therefore changes the classification of Assessable Property for such Lot. The applicable Assessment for such a Lot shall be prorated according to the rate required for each type of ownership.
- 4.03 Rental Property. The Association's Board of Directors shall develop, publish, and enforce regulations for reclassifying residential property within Eaglesgate that will be leased or rented. The purpose of these rules is to protect the health, safety, property value, and quality of life for all residents; provide for full inclusion of tenants in the community life; and insure that community facilities and physical features are not overburdened. Tenants will have access to common property and other benefits offered to other residents. There is nothing in this provision which prohibits tenants from using common areas. Tenants are given this right under subparagraph (h).

- (a) An Eaglesgate Living Unit shall be classified as rental property when owned by an absentee Owner and occupied by "residential tenants" as defined in Section 1.27. The Association Board may charge a uniform Assessment for such property different than that charged to Owners who occupy their residences in the community. Further, the Board shall develop procedures for setting and collecting annual dues for such property.
- (b) As provided herein, an Owner leasing or renting property within the community must register such property with the Association in writing prior to tenant occupation.
- (c) Absentee Owners of rented Living Units shall be responsible for the compliance of their tenants with all provisions of the Declaration of Covenants and shall be subject to the same enforcement policies as resident Owners. Property owners found in violation of these provisions will be subject to fines and penalties as determined by the Board of Directors. Failure to pay those penalties and associated enforcement and legal costs may result in a lien on the property.
- (d) The dues for property where there is a lease-to-purchase contract will be the current rental property dues until the property is transferred to the new Owner. Pro-rated residential dues will be paid by the new resident homeowner as of the first of the month following closing.
- (e) The Board shall develop and publish standards and requirements regarding the minimum square feet of living space available per occupant of a rental household.
- (f) An absentee Owner of a residence where a former spouse and/or children live may avoid classification as rental property by certifying in writing to the Board that the tenants are not paying rent.
- (g) A resident Owner who has to move out of the community before selling the home may rent that home to tenants for up to one year without having to pay the rental property dues.
- (h) Lawful tenants occupying residences upon a Lot within this community shall have the right to use common areas and amenities as other residents.
- 4.04 <u>Billing of Annual Assessments</u>. At such time or times as the Association Board may determine, the Association shall levy the Annual Assessment. The Association shall send a written bill to each Owner stating the amount of the Annual Assessment imposed against each Lot which is Assessable Property owned by the Owner, the time period for payment thereof, and the interest rate to be charged for late payments thereof. Each Annual Assessment shall be due and payable on a date established by the Association Board and shall become delinquent on a date established by the Association Board. The Association Board may establish payment procedures to allow payment of

the Annual Assessment in increments during the year the Assessment is made, provided that this privilege is extended to all Owners on an equal basis, and provided that reasonable notice is given of each payment date, of the interest to be charged for late payments, of the liens established by this Declaration, and of the suspension of membership rights as a consequence of the failure to pay. The initial assessment shall be due upon occupancy of the house.

- Commencement of Assessments. With regard to the Property, and any Annexation Property, each part of the Assessable Property shall become subject to the Assessments set forth herein on the first day of the fiscal quarter following the month in which each part meets the definition of a Lot. Such Assessments shall be adjusted and prorated according to the number of quarters remaining in the fiscal year of the Association as such fiscal year is set forth in the Association By- Laws. Annual dues assessed are determined by the classification of the residence as of January 1 of each calendar year. Whether classified as residential property or rental property, all pro-rated dues are matters to be settled between the purchaser and the seller prior to the closing on the property.
- Late Payments or Non-Payment. Each Owner shall be deemed to covenant and agree to pay to the Association the assessments provided for herein, and each agrees to the enforcement and collection of the assessments in the manner herein specified. In the event the Association employs an attorney or attorneys for collection of any assessment, whether by suit or otherwise, or to enforce compliance with or specific performance of the terms and conditions of this Declaration, or for any other purpose in connection with the breach of this Declaration, each Owner agrees to pay all expenses, including collection costs, administrative costs, and reasonable attorneys' fees, together with such late charges as provided herein, in addition to any other amounts due or any other relief or remedy obtained against said Owner. In addition to any other remedies herein or by law provided, the Association may enforce each such obligation in any manner provided by law or in equity, or, without any limitation by the foregoing, by any or all of the procedures provided below.
 - (a) The Association may from time to time establish or change the rate of interest which shall be charged for the payment after the delinquency date of any portion of an Assessment. In the event of a default in payment of any such assessment when due, the assessment shall be deemed delinquent, and shall bear interest at the rate established by the Association's Board of Directors, not to exceed 18 percent and calculated on a per day basis.
 - (b) In the event of default in the payment of any one (1) or more installments of the Annual Assessment established hereunder, the Association may declare any remaining balance of said Annual Assessment at once due and payable.
 - (c) In the event that an Owner shall fail to fully pay the Assessment by the delinquency date thereof, such unpaid amount shall become a binding personal obligation of such Owner, and the Association shall have the right, pursuant to the provisions of Section 15.03 hereof, to enforce the lien for Assessments imposed by Section 4.01. The Association shall have the right and duty to take all appropriate actions and steps to collect any such unpaid

Assessments. Each delinquency shall constitute a separate basis for a demand or claim of lien or liens, but any number of defaults may be included within a single demand or claim of lien or liens on account of prior delinquencies and shall be deemed to include subsequent delinquencies and amounts due on account thereof. The Association may institute a suit to recover a money judgment for the same, together with interest thereon and reasonable expenses of collection, including attorney fees, without foreclosing or waiving the lien hereinbefore provided.

- (d) Upon demand by an Owner, the Association shall issue and furnish to such Owner, within a reasonable period of time, a written certificate stating that all Assessments, including interest and costs, have been paid with respect to any specific Lot owned by said Owner, as of the date of such certificate. If all Assessments have not been paid, the certificate may set forth the amount then due and payable. The Association may make a reasonable charge for the issuance of such certificate. Any such certificate, when duly issued as herein provided, shall be conclusive and binding with regard to any matter herein stated as between the Association and any bona fide purchaser or encumbrancer of the Lot in question.
- 4.07 <u>Enforcement Authority</u>. The Association may initiate or enter into any enforcement action regarding non-payment or delinquent payment of assessments provided by law or in equity. Those actions include, but are not limited to, the following.
 - (a) Enforcement by Suit. The Association may cause a suit at law to be commenced and maintained in the name of the Association against any Owner to enforce each assessment obligation. Any judgment rendered in any such action shall include the amount of the delinquency, together with interest thereon at the rate set by the Association from the date of delinquency, court costs, costs of collection, administrative costs, and reasonable attorneys' fees in such amount as the court may adjudge against the delinquent Owner.
 - Enforcement by Lien. To the full extent of the law, there is hereby created a claim of lien, with power of sale, on each and every Lot within Eaglesgate. The purpose of such claim is to secure payment to the Association of any and all assessments levied against any and all Owners of such lots under these Covenants, together with interest thereon at the rate established by the Association from the date of delinquency, and all costs of collection which may be paid or incurred by the Association in connection therewith, including reasonable administrative and attorneys' fees. At any time within fifteen (15) days after the occurrence of any default in the payment of any such assessment, the Association or any authorized representative shall mail a written demand for payment to the defaulting Owner, on behalf of the Association. Each default shall constitute a separate basis for a demand or

- claim of lien or liens, but any number of defaults may be included within a single demand or claim or lien.
- (c) <u>Election to File as Lien</u>. If such delinquency is not paid within ten (10) days after delivery of such demand, or even without such a written demand being made, the Association may elect to file such a claim or lien on behalf of the Association against the Lot of the defaulting Owner. Such a claim of lien shall be executed and acknowledged by any officer of the Association. Section 15.03 provides details on the content of such liens.
- Attachment. Upon recordation of a duly executed original or copy of such a (d) claim or lien, the lien shall immediately attach and become effective in favor of the Association as a lien upon the property against which such assessment was levied. Such a lien shall have priority over all liens or claims created subsequent to the recordation of the claim of lien thereof. Any such lien may be foreclosed by appropriate action in court or in the manner provided by law for the foreclosure of a deed of trust, with a power of sale, as set forth by the laws of the State of Missouri, as the same may be changed or amended. The lien provided for herein shall be in favor of the Association and shall be for the benefit of the Members. The Association may acquire, hold, lease, mortgage, and convey any such Lot. In the event such foreclosure is by action in court, reasonable attorneys' fees, court costs, title search fees, interest and all other costs and expenses shall be allowed to the Association to the extent permitted by law. Each Owner, by becoming an Owner in Eaglesgate, hereby expressly waives any objection to the enforcement and foreclosure of this lien in this manner.
- (e) Preparation Fees. In addition to the other fees and expenses owed by the defaulting Owner as provided herein, the defaulting Owner shall pay a lien preparation fee in the amount of \$100.00, and if the lien is subsequently released, an additional lien release preparation fee of \$50.00, together with all costs incurred by the Association with regard to said lien. These fees are in addition to attorneys' fees. The Board of Directors may increase or decrease these fees, or add others, as circumstances warrant.
- (f) Necessary and Proper. The Association delegates this enforcement authority to the Board of Directors. The Board shall have all power necessary to enforce the rights of the Association concerning the non-payment or delinquent payment of assessments as provided by law or equity. The Board shall have the authority to decide all matters concerning such assessments, including whether to seek enforcement and how those assessments should be tried before a court of law if paid, settled, or otherwise resolved.

4.08 <u>User Fees and Charges</u>.

- (a) In addition to the Annual Assessments, the Association Board may levy and collect charges and fees from all Owners for the use of Common Property for the purpose of maintaining, refurbishing, replacing and repairing Common Property and the Common Property Improvements, and operating services on Common Property.
- (b) In establishing user fees and charges, The Association Board may formulate reasonable classifications of users. Fees and charges shall be uniform within each classification, but need not be uniform form classification to classification.
- (c) If any Owner shall fail to pay any user fee or charge when due and payable, the Association Board may immediately suspend such Owner's right of enjoyment of the Common Property or services thereon except as to Private Open Space and the Retention Pond and may take whatever action deems necessary to enforce such suspension.
- 4.09 <u>Additional Procedures</u>. The Association Board shall have the right to adopt procedures for the purpose for making the Assessments, user fees and charges provided for herein and for the billing and collection of the same, provided that such procedures are not inconsistent with the provisions hereof.

4.10 Special Assessments.

- (a) In addition to the Annual Assessments authorized by Section 4.01 hereof, the Association may levy in any year a Special Assessment, applicable to that year only, for the purpose of defraying, in whole or in part, the cost of any construction or reconstruction, repair or replacement of Common Property Improvements including any capital improvement upon the Common Property, or the cost of any utility deemed necessary by the Association Board to serve the Property including the necessary fixtures and personal property related thereto, or any unexpected cost or expense of the Association, as the Association Board may determine.
- (b) A Special Assessment shall become effective upon written notice by the Association Board. Billing of Special Assessments shall be handled according to the procedures set out in Section 4.03 hereof.
- (c) The Association must present a proposed Special Assessment to the Members at a meeting to be called for that purpose pursuant to Section 3.06 hereof. The Special Assessment will then be submitted to the residents by written ballot and shall be deemed duly approved by the Members if a majority of the

Members then allowed and eligible to vote, pursuant to Section 3.07(e), approves such Special Assessment.

(d) Any Lots owned by builders shall not be subject to Special Assessments.

ARTICLE V

Use of Funds

- 5.01 <u>Purposes of Which Funds May Be Used</u>. The Association shall apply all funds received by it pursuant to this Declaration and all other funds and property received by the Association, including the proceeds of loans referred to in Sections 5.02 and 5.05 and the accumulated funds referred to in Section 5.03, to the following:
 - (a) the operating costs and expenses of the Association, including planning and implementation of the community programs;
 - (b) the planning, design, acquisition, improvement, construction, maintenance and equipping of Common Property Improvements;
 - (c) Association programs and services conducted on or in Common Property;
 - (d) Payment of all principal and interest when due on all loans made to the Association to the extent required under any agreement with Note Holders pursuant to Section 5.02 hereof;
 - (e) Payment of all real and personal property taxes and assessments, if any, separately levied upon or assessed against the Association or any property owned by the Association;
 - (f) Payment of all premiums and charges for all policies of insurance or surety bonds, as deemed by the Association Board to be necessary and appropriate, including but not limited to workmen's compensation insurance, to the extent necessary to comply with any applicable law and then-current insurance practices, and indemnity, faithful performance, fidelity and other bonds as the Association Board shall deem appropriate, or required to carry out the Association functions or to insure the Association against any loss from malfeasance or dishonesty of any employee or other person charged with the management or possession of any Association funds or property; and
 - (g) The repair, improvements, construction, operation or extension of any utility servicing the Property or any utility deemed reasonably necessary by the Association Board to service the Property.

- 5.02 <u>Handling of Funds</u>. In order to secure the repayment of any and all sums borrowed by it from time to time, the Association Board is hereby granted the right and power:
 - (a) to assign and pledge revenues received and to be received by it under any provision of this Declaration, including, but not limited to, the proceeds of the Assessments payable hereunder; and
 - (b) to enter into agreements with Note Holders with respect to the collection and disbursements of funds, including, but not limited to, agreements wherein the Association covenants:
 - (i) to assess the Assessments on a given day in each year and subject to the limitations specified in Article IV;
 - (ii) to establish sinking funds or other security deposits, or both;
 - (iii) to apply funds received by the Association to the payment of all principal and interest when due on such loans or to apply the same to such purpose after providing for costs of collection;
 - (iv) to establish such procedures as may be required by the Note Holders, but not inconsistent with the Declaration;
 - (v) to provide for the custody and safeguarding of all funds by the Association; and
 - (vi) to negotiate and arrange the amount, terms, and rate or rates of all borrowing and the provisions of all agreements with Note Holders.
- 5.03 <u>Accumulation of Funds Permitted.</u> The Association shall not be obligated to spend in any calendar year all the sums collected in such year by way of annual assessments, or otherwise, and may carry forward, as surplus, any balances remaining; nor shall the Association be obligated to apply such surplus to the reduction of the amount of the Annual Assessment in the succeeding year, but may carry forward from year to year such surplus as the Association Board may determine to be necessary or desirable for the greater financial security of the Association and the effectuation of its purposes, including accruing funds for the replacement of facilities.
- 5.04 <u>Posting of Bond</u>. The Association, acting through the Association Board, shall require that all persons or entities who handle the Association funds or monies, which funds and monies shall be deposited in federally insured banks or savings and loans, post bonds sufficient in amount to indemnify the Association from any loss.
- 5.05 <u>Mortgaging of Common Property</u>. Except as set forth in this Section, and subject to the approval of any holder of an existing lien on the Common Property (the "Developer Loan Lien"), the Association may mortgage any Common Property to which it has clear title; provided, however, that any such mortgage shall be subject to the approval of two-third (2/3) of the Members

who are present in person or by proxy and voting in a duly constituted Association election or meeting under the provisions of Section 3.07. The Association shall not mortgage any Common Property to the Developer, or to any other entity or person to secure any conveyance, loan or advance made to the Association by the Developer. The Developer shall not take any action, the result of which may subject any Common Property to a judgment lien or otherwise jeopardize any Common Property, to satisfy a debt of the Developer.

ARTICLE VI

Common Property

6.01 Use of Common Property.

- (a) Every Owner, by reason of such ownership shall have a right and easement of enjoyment in and to all Common Property, and such easement shall be appurtenant to and shall pass with every Lot upon transfer. All Residential Tenants who are not also Owners shall have a nontransferable privilege to use and enjoy all Common Property for so long as they are a Resident and the Owner has sent the Board a letter approving of that privilege. Any guest of a Member shall be entitled to a right or privilege of enjoyment of Common Property subject to such regulations as may be promulgated by the Association Board. Each such guest shall be accompanied by the Member sponsoring such guest at all times such guest is using the Common Property.
- (b) All such rights, easements and privileges conferred under this Article VI shall, however, be subject to the right of the Association Board to:
 - (i) establish, adopt, promulgate, amend and rescind reasonable rules and regulations pertaining to the use, operation and maintenance of Common Property which shall enhance the preservation of such facilities, promote the safety and convenience of the users thereof, and which shall serve to promote the best interests of the Members and the Eaglesgate community;
 - (ii) determine which, if any, Common Property may be used and enjoyed by, or conveyed or dedicated to the general public or a federal, state, or local government body; provided, however, that Property shall not be conveyed to a public body unless the Association Board has obtained the prior approval of two-thirds (2/3) of the Members who are present, in person or by proxy, and voting at an Association meeting at which a quorum is present or voting in a referendum called for such purpose after proper notice is given and according to the provisions of Section 3.07;

- (iii) levy user fees and charges pursuant to Section 4.08 of this Declaration and to charge reasonable admissions or other charges or fees for the use of any recreational facility;
- (iv) borrow money for the purpose of acquiring, developing or improving any Common Property including improvements thereon, and in aid thereof to mortgage the same, provided that the rights of any such mortgagee shall be subordinate to the rights, easements and privileges herein granted and assured; provided, further, that any such mortgage shall be subject to the approval of two-thirds (2/3) of the Members who are present in person or by proxy and voting at an Association meeting at which a quorum is present or voting in a referendum called for such purpose after the proper notice is given under the provisions of Section 3.07; and
- (v) apply for, accept and expend loans or grants from federal, state or local governments and to comply with any conditions required by such governments in order to obtain such loans or grants including conditions relating to the use and enjoyment of Common Property by the general public.
- Damage or Destruction of Common Property by Owner. In the event any Common Property is damaged or destroyed by an Owner, Resident or any of their guests, tenants, licensees, agents or members of their families, such Owner and Resident do hereby authorize the Association to repair such damaged areas. The Association shall repair such damaged area in a good and workmanlike manner in conformance with the original plans and specifications of the area involved, or as the area may have been modified or altered subsequently by the Association at the discretion of the Association. The amount necessary and actually expended for such repairs shall be a Special Assessment upon the Lot or Living Unit of said Owner, shall be a lien upon the Lot or Living Unit of said Owner and shall be enforceable as other Assessments under Article IV.
- 6.03 <u>Maintenance of Common Property</u>. The Association shall maintain the Common Property according to at least the same standard of maintenance required of Owners.
- 6.04 <u>Suspension of Rights</u>. Except with regard to the Private Open Space and Retention Pond, the Association shall have the right to suspend the right or privilege under this Article VI of any Member for any period during which the Assessments or user fees and charges assessed under Article IV hereof remain delinquent, and may suspend said right or privilege in connection with the enforcement of any rules and regulations relating to Common Property in accordance with the provisions of this Article VI.

ARTICLE VII

Designation of Use of Property

7.01 <u>Change of Designation</u>. The Association may not change the designation of a Lot without the mutual consent of the Owner thereof and the DRC, together with such additional government approval as may be required. Any change in the designation of Common Property shall require the approval of three-fifths (3/5) of the Association Board.

ARTICLE VIII

Design Review Committee

8.01 <u>Purpose, Powers and Duties of the DRC.</u> The purpose of the DRC is to assure that all proposed uses and any construction or alteration of any Structure which take place on any Lot or any other Property shall be performed in conformity with the objective of high quality environmental design and development. To carry out that purpose, the DRC shall have all of the rights, powers and duties conferred upon it pursuant to the provisions of this Article VIII, including the right to approve any and all proposed uses, site plans and Structures to be constructed on the Property, including the Common Property Improvements, except that the DRC shall not have the right, without the approval of the Association Board, to disapprove a use for a Lot which is within the use category designated for such Lot. The DRC shall also have the right to approve or disapprove any and all proposed external alterations or use changes for Lots, Living Units or Common Property.

8.02 Composition and Appointment.

- (a) The DRC shall be comprised of five (5) members, each serving a two (2) year terms. Members of the Committee shall be appointed by a majority vote of the Association Board of Directors. They may be re-appointed for one additional term. If a person has been off the Committee for at least one term, that person may be appointed again as a new member. All members of the Committee shall be homeowners residing in Eaglesgate. When possible, the Board will have as members of the Committee at least one architect registered in Missouri and one civil engineer licensed in Missouri.
- (b) Any DRC member may resign at any time by giving written notice of such resignation to the Chairman of the DRC. Such resignation shall take effect on receipt thereof by the Chairman. Any member of the DRC may be removed with or without cause by the DRC or the Association Board. If the Association Board, as appropriate, fails to act to fill a vacancy within thirty (30) days of the receipt of a written notice or resignation, a majority of the remaining members of the DRC shall appoint a new member to fill the vacancy.

(c) If any vacancy shall occur in the membership of the DRC by reason of death, resignation, removal, or other reason, Eaglesgate residents will be notified by community notice and invited to submit their names as candidates for the vacancy by emailing drceaglesgate@gmail.com. From those submissions, the DRC shall nominate candidates for the vacancy or vacancies to the Board. A majority of the Board shall appoint someone from the DRC nominations to fill any vacancy on the Committee

8.03 Officers, Subcommittees, and Reimbursement.

- (a) Members of the DRC shall appoint a Chairman from among their number and may appoint from among their number such other officers and subcommittees as they shall from time to time determine.
- (b) In addition, the DRC may establish advisory committees, which may include such members of the DRC, employees or Members of the Association, or other persons as the DRC may in its discretion determine. Any such subcommittees or advisory committees may make recommendations to the DRC but shall have no power to finally exercise any of the power and authority of the DRC under this Declaration.
- (c) Members of the DRC and advisory committees may receive reimbursement for expenses as the Association Board may determine for out-of-pocket costs incurred in the performance of their duties as a member of the DRC or advisory committees.

8.04 Operations of the DRC.

- (a) Meetings. The DRC shall hold regular meetings as may be determined by the members of the DRC. Special meetings of the DRC may be called by the Chairman of the DRC upon the written request of a majority of the members of the DRC then in office. Regular and special meetings of the DRC shall be held at such time and at such place as the members of the DRC shall specify.
 - (i) Notice of regular and special meetings need not specify the purpose or purposes for which the meeting is called. Attendance of a member of the DRC at a meeting shall constitute a waiver of notice of such meeting.
 - (ii) At each meeting of the DRC, the presence of a majority of the members then in office shall constitute a "quorum," herein so called, for the transaction of business. Except as otherwise provided herein, the act of such majority of members of the DRC present at any regular or special meeting thereof at which a quorum is present shall constitute the act of the DRC.

- (iii) In the absence of a quorum, a majority of the members of the DRC present at the time and place of the meeting may adjourn the meeting from time to time and until such time as a quorum shall be present. At any adjourned meeting at which a quorum is present, any business may be transacted which might have been transacted at the meeting as originally called.
- (iv) The DRC shall maintain both a written record of votes and minutes for each of its meetings. Upon written request, the DRC shall make such records and minutes and current copies of its Development Guidelines available at reasonable places and times for inspection by Members and prospective Members of the Association.
- (b) <u>Activities</u>. The DRC shall adopt and promulgate and, as it deems appropriate, amend, the Development Guidelines as provided in Section 8.06 hereof and will, as required, make findings, determinations, rulings, and orders with respect to the conformity with said Development Guidelines of plans and specifications to be submitted for approval to the DRC as provided in Section 8.07 hereof.
 - (i) As required, the DRC shall issue permits, authorizations, or approval pursuant to the directions and authorizations contained herein.
 - (ii) Any three (3) or more of the members of the DRC may be authorized by the DRC to exercise the full authority of the DRC with respect to the review of plans and specifications pursuant to the provisions of the Article and with respect to all other matters as may be specified by resolution of the DRC, except with respect to the adoption or promulgation of the Development Guidelines.
 - (iii) The unanimous action of three (3) or more members of the DRC in issuing an approval based upon specified conditions or a modification or any plans and specifications submitted under the provisions of this Article, or in issuing an approval or disapproval of any permit or authorization, shall be final and binding upon the DRC, subject, however, to an appeal to the DRC as a whole
 - (iv) Anyone applying for such approval, permit or authorization may file a written request to have the matter in question reviewed by the entire DRC, within thirty (30) days after receipt of notice of any decision that the applicant deems to be unsatisfactory. Upon filing of any such request, the matter with respect to which such request was filed shall be submitted to, and reviewed as soon as possible, by the entire DRC. Thereafter, the unanimous decision of all members of the DRC with respect to such matter shall be final and binding.

8.05 <u>Development Guidelines.</u>

- (a) As contemplated by and pursuant to the provisions of this Article VIII, the DRC may adopt, promulgate, amend, revoke and enforce design and development guidelines, hereafter referred to as the Development Guidelines, for the purposes of:
 - (i) governing the form and content of plans and specifications to be submitted for approval pursuant to the provisions of Section 8.07;
 - (ii) governing the procedure for such submission of plans and specifications; and
 - (iii) establishing policies, requirements, standards, restrictions and specifications with respect to the approval and disapproval of all proposed uses and with respect to all construction or alteration of any Structure on any Lot, Living Unit, Easement Area or Common Property.
- (b) The DRC shall make a published copy of its current Development Guidelines readily available to Members and prospective Members of the Association and builders which Guidelines shall include the Residential Plan Submission Procedure, Site Planning and Landscape Standards, and Residential Design Standards.

8.06 Submission of Plans and Specifications.

- (a) No Structure shall be commenced, erected, placed, moved onto or permitted to remain on any Lot, nor shall any existing Structure upon any Lot be altered in any way which materially changes the exterior appearance thereof, nor shall any new use be commenced, unless plans and specifications (including a description of any proposed new use) therefore shall have been submitted to and approved in writing by the DRC.
- (b) Home construction and remodeling requests submitted to the DRC shall be in such form and shall contain such information as may be required by the DRC in the Development Guidelines promulgated by the DRC pursuant to Section 8.06 of this Declaration.
- (c) At a minimum, submissions to the DRC shall include (in such detail and according to the scale as required by the Development Guidelines) five (5) copies, either paper or electronic, of:
 - (i) the site plan;
 - (ii) floor plans;

- (iii) building materials;
- (iv) landscape plan;
- (v) exterior color scheme.
- (d) Such submissions shall include, among the other requirements of the Development Guidelines, a site plan showing building location and relation to the lot lines with finish grades; floor plans showing not more than two (2) exterior elevations per page, noting the location of all windows, doors, openings, and finish materials, together with samples, pictures, or links to materials and colors,; landscaping plans, noting types of shrubs and trees; and any other information requested on the construction request form provided by the DRC.
- 8.07 Approval of Plans and Specifications. Upon approval by the DRC of any plans and specifications submitted hereunder, a copy of such plans and specifications, as approved, shall be deposited as a permanent record with the DRC and a copy of such plans and specifications bearing such approval in writing shall be returned to the applicant submitting the same. Approval for use in connection with any Lot or Structure of any plans and specifications shall not be deemed a waiver of the DRC's right, in its discretion, to disapprove similar plans and specifications or any of the features or elements included therein if such plans, specifications, features or elements are subsequently submitted for use in connection with any other Lot or Structure. Approval of any such plans and specifications relating to any Lot or Structure, however, shall be final as to that Lot and such approval may not be revoked or rescinded thereafter, provided that there has been adherence to, and compliance with, such plans and specifications, as approved, and any conditions attached to any such approval. The DRC, in its discretion, is permitted to approve deviations from the Development Guidelines and from this Declaration when, in its judgment, such deviations will result in a more commonly beneficial use. Such approval must be granted in writing and when the DRC approves and grants a deviation from this Declaration, such approved deviation shall for all purposes amend this Declaration but only to the limited extent of such specifically approved deviation. No approved deviation shall be deemed to act as a precedent in respect of any other requests for approvals of deviations.

8.08 <u>Disapproval of Plans and Specifications.</u>

- (a) The DRC shall have the right to disapprove any plans and specifications submitted hereunder as determined by the DRC in its sole judgment and discretion for any reason including, among others:
 - (i) failure to include information is such plans and specifications as may have been requested by the DRC or as required herein or in the Development Guidelines;

- (ii) failure of such plans or specifications to comply with this Declaration or the Development Guidelines;
- (iii) objection to the exterior design, appearance or materials of any proposed Structure or improvements;
- (iv) incompatibility of any proposed Structure, uses or Lot improvements to other existing neighboring Structures;
- (v) objection to the site plan due to incompatibility with neighboring Lots;
- (vi) objection to the grading plans and drainage patterns for any Lot;
- (vii) objection to the color scheme, finish, proportions, style or architecture, height, bulk, safety or appropriateness of any proposed Structure or improvement;
- (viii) failure to satisfy minimum or maximum floor area requirements or standards:
- (ix) objections to the parking areas proposed for any Lot based on among other things:
 - (A) incompatibility with proposed uses and Structure of the Lot;
 - (B) insufficiency of size or the parking area in relation to the proposed use; or
 - (C) undesirable alteration of the flow of water over or through the Lot;
- (x) any other matter with respect to such submitted plans and specifications, whether or not such matters are included in the Development Guidelines, if such matters, in the sole judgment and discretion of the DRC, would conflict with the value and amenities of the Eaglesgate community established in the Development Plan and as may be set forth in this Declaration or the Development Guidelines;
- (xi) any other matter which, in the judgment of the DRC, would render a proposed Structure or use in harmonious with the standards for Eaglesgate as set forth in the Development Plan.
- (b) In any case in which the DRC shall disapprove any plans and specifications submitted hereunder, or shall approve the same only as modified or upon

- specified conditions, such disapproval or qualified approval shall be accompanied by a statement of the grounds upon which such action was based. In any such case the DRC shall, if requested, make reasonable efforts to assist and advise the applicant in order that an acceptable proposal may be prepared and submitted for approval.
- (c) The approval or disapproval of plans and specifications by the DRC shall not be constructed as approval or disapproval of engineering decisions or of compliance of such plans and specifications with zoning and building ordnances nor with any such industry or governmental standards, rules, regulations and codes. It is the responsibility of each Owner to employ properly qualified and, as applicable, licensed professionals to design and construct this residence. By approving or disapproving the plans and specifications neither the DRC nor any member thereof assumes any liability or responsibility therefore or for any defect in any Structure or part thereof constructed from such plans and specifications.
- 8.09 <u>Failure to Act</u>. In the event that the DRC shall fail to take action on any plans and specifications as herein provided within thirty (30) business days after receipt of "preliminary" plans and specifications, together with the fees authorized by Section 8.13 hereof, and such other requirements as called for by the Development Guidelines, the same shall be deemed to have been approved as submitted, and no further action by the DRC shall be required for the applicant to begin construction. Such approval shall be placed in writing on the plans and specifications and shall be returned to the applicant.
- 8.10 <u>Inspection Rights</u>. After reasonable notice and at any reasonable time or times, any agent of the Association or the DRC may enter upon any Lot for the purpose of ascertaining whether the use or maintenance of such Lot or the construction of any Structure thereon is in compliance with the provisions hereof. Neither the Association, nor the DRC, nor any agent thereof shall be deemed to have committed a trespass or, other wrongful act solely by reason of such entry or inspection, provided such inspection is carried out in accordance with the terms of this Section.
- 8.11 <u>Violations</u>. If any Structure shall be erected, placed, maintained or altered upon any Lot, or any new use commenced on any Lot, otherwise than in accordance with the plans and specifications approved by the DRC pursuant to this Declaration, such erection, placement, maintenance or alteration shall be deemed to have been undertaken in violation of this Article and without the approval required herein. If in the opinion of the DRC, such violation shall have occurred, the DRC shall notify the Association. If the Association Board shall agree with the determination of the DRC with respect to the violation, then upon written notice of the violation to the Owner from the Association Board (which shall be deemed to have been delivered if sent by certified or registered mail, return receipt requested, postage paid), any such Structure so erected, placed, maintained or altered upon any Lot in violation hereof shall be removed or altered, and any such use shall be terminated, so as to extinguish such violation. If the Owner of the Lot upon which such violation exists shall not have taken reasonable steps toward the removal or termination of the same within the time specified in such notice, the Association shall have the right to pursue its Right of Action as provided in Article XV hereof together with all remedies whether at law or in equity

and whether specified herein or in Article XV hereof, and including but not limited to the remedy of injunctive relief and obtaining a monetary judgment for all costs, expenses, including reasonable attorney's fees, and damages.

- 8.12 Fees. As a means of defraying its expenses, the DRC may, in unusual circumstances, charge and collect a reasonable and appropriate fee as established from time to time and published in the Development Guidelines. The fees shall be made payable to the Association and shall be payable at the time plans and specifications are submitted as a condition precedent for the review and approval of such plans and specifications. Such fees shall not be charged for ordinary review of plans and no fees shall be charged without prior notice to the builder.
- 8.13 <u>Non-discrimination by DRC</u>. The DRC shall not discriminate against any applicant requesting its approval of plans and specifications because of such applicant's race, color, sex, religion, national origin, family composition or marital status. Further, the DRC in the exercise of its power granted pursuant to this Declaration shall not take any action which is intended to or does, in effect, discriminate against persons of a particular race, color, sex, religion, national origin, family composition or marital status.

ARTICLE IX

Easements

9.01 <u>Easements</u>.

- (a) In respect of the Easement Area of each Lot and the Common Property, the Association reserves to itself, its assignees, and designees the right, power, and authority to
 - (i) erect, install, construct, and maintain wires, lines, conduits, and poles and the necessary or proper attachments and appurtenant structures in connection with the transmission of electricity, telephone, fire alarm systems, communication systems, television cables, and other utilities and similar facilities;
 - (ii) erect, install, construct, and maintain storm water drains, land drains, public and private sewers; pipe lines for supplying gas, water, and heat; and of any other public or quasi-public facility, service or function, and appurtenant structures, whether aboveground or underground
 - (iii) control slope, including the right to grade and plant slopes and prevent any activity which might interfere with slope ratios approved by the Association and DRC or which might create erosion or sliding problems or which might change, obstruct, or retard drainage flow;

- this section 9.01(a) shall not be construed to exempt Structures erected or placed in the Easement Area from the provisions of Article VIII.
- (b) The Association reserves unto itself, its assignees, successors, and designees, the right, power and authority to direct and control the installation of facilities, in cooperation with a public authority or any utility company, which will install, own, operate and maintain the respective facilities, which utilities and drainage services (as provided for in paragraphs (a) (i), (ii) and (iii) of this Section 9.01) shall be installed in and occupy any specific easement. Within any easements, no Structure, planting or other material or improvement shall be placed or permitted to remain which may damage or interfere with the installation and maintenance of utilities, or which may change the directional flow of drainage channels in the easement, or which may alter, obstruct or retard the flow of water through drainage channels within the Easement Areas, or which may change or prevent the intended use of any easement.
- (c) Subject to all of the other Restrictions contained in this Declaration, and subject to the easements and rights thereto pursuant to the Plat, each Owner shall have the right to use the Easement Area of his Lot in any manner not inconsistent with the purposes for which such Easement Area is reserved, and the area within any Easement Area and all improvements within the bounds of such Easement Area shall be maintained continuously by the Owner except for such improvements for which a public authority or utility company is or may become responsible for maintenance.
- (d) Notwithstanding anything herein to the contrary, each Owner covenants and agrees that, in cooperation with the Association, each Owner shall execute all grants of easements, grants of right-of-way or any other similar grant or conveyance documentation required to be executed by an Owner in order to grant and convey to any public authority or utility company, their assigns or lessees, the right, privilege, and easement to lay, construct, maintain, alter, inspect, repair, replace, protect, relocate, change the size of, operate and remove all utility lines, service taps, distribution facilities, valves, regulators and other equipment appurtenant to and necessary for providing any and all of the utility and drainage services as provided for in paragraphs (a) (i), (ii) and (iii) of this Section 9.01.
- 9.02 Entry. The Association reserves to itself, its agents, designees, committees, successors, and assigns the right at all reasonable times and upon reasonable notice to enter upon all parts of the Easement Areas of each Lot for any of the purposes for which said easements or right-of-ways are reserved, without being deemed to have committed a trespass or wrongful act solely by reason of such entry. The Association, its agents, designees, committees, successors, and assigns shall be responsible for leaving each Easement Area in good repair and condition following any work or activity within such Easement Area pursuant to the provisions of Section 9.01.

9.03 <u>Conveying an Easement Area.</u> The Association may convey an Easement Area to a public authority or utility company where such conveyance is required by the public authority or utility company as a prerequisite to installing the utility facility on the Easement Area or where such conveyance is required by the public authority or utility company as a prerequisite to accepting ownership of the utility facility for operation and maintenance.

ARTICLE X

General Restrictions

10.01 Maintenance Required by Owner.

- Each Owner shall keep all of his Lots (including Easement Areas, if any), and (a) all improvements therein or thereon, in good order and repair, including, but not by way of limitation, the seeding, watering and mowing of all lawns, the pruning and cutting of all trees and shrubbery and the painting (or other appropriate external care) of all buildings and other improvements, all in a manner and with such frequency as is consistent with safety and good property management. There is reserved to the Association, its agents, successors, designees or assigns a "maintenance easement" on Property lying between the foundation of any Structure on any Lot and the property line of said Lot to permit the Association, its agents, successors or assigns, at its election, to maintain said Property at any reasonable hour. The Association shall have the right, after written notice to the Owner of the affected Lot or Living Unit as hereinafter provided, to remove trash or rubbish and to cut grass, weeds and vegetation and to trim or prune any hedge or other planting that, in the opinion of the DRC, by reason of its location or height or the manner in which it is permitted to grow, is detrimental to adjoining Lots or Property or is unattractive in appearance. The Association shall further have the right to care for vacant and unimproved Property and to remove grass, weeds and rubbish therefrom and to any and all things necessary or desirable in the opinion of the DRC to keep such Property in neat and good order, all at the cost and expense of the Owner. Such cost and expenses incurred by the Association shall be paid to the Association upon demand and, if not paid within ten (10) days thereof, shall become a lien upon the Property affected, equal in priority to the liens provided for in Article IV hereof.
- (b) The DRC shall give fifteen (15) days' written notice to the Owner in violation of this Restriction, setting forth the specific violation or breach of this Restriction and the action required to be taken by the Owner to remedy such violation or breach; if, at the end of such time, reasonable steps to accomplish such action have not been taken by the Owner, the DRC may pursue its Right of Action and shall have such other remedies at law or in equity as may then exist or a provided in Article XV hereof.

- 10.02 <u>Land Use and Building Type</u>. Lots shown on the Eaglesgate Plat, shall be used solely for residential purposes. The term "residential purposes" as herein, excludes hospitals, clinics, hotels, industrial, commercial and professional uses, whether from homes, residences or otherwise, and all such uses of the Lots are expressly prohibited. No building shall be erected, altered, placed or permitted to remain on any Lot unless it is an approved Structure and no previously approved Structure shall be used for any purpose other than that for which it was originally approved. No Lot shall be split, divided or subdivided for sale, resale, gift, transfer or otherwise without the written consent of the Association Board.
- Landscape Restrictions. No tree or shrubbery shall be maintained in such a manner 10.03 as to obstruct the view of vehicular traffic. No tree having a diameter of four (4) inches or more (measured from a point two (2) feet above ground level) shall be removed from any Lot without the express written authorization of the DRC. The DRC may adopt and promulgate rules and regulations regarding the requirement of planting trees, preservation of trees and other natural resources and wildlife to protect and encourage the preservation of the ecological balance of the Property. The DRC may mark certain trees, regardless of size, as not removable without written authorization. Landscaping must conform to the area in the opinion of the DRC. Each Lot must be sodded with fine leaved turf-type tall fescues, ryes and blue grasses, except in areas to be left in a natural state upon approval of the DRC. In carrying out the provisions of this Section 10.03, the DRC, the Association and its agents or designees may come upon any Lot (following reasonable notice) during reasonable hours for the purpose of inspecting and marking trees. Notwithstanding the foregoing, the DRC, in its sole discretion may grant written permission to an Owner to partially or completely grass seed a Lot in lieu of sodding. The grass seed and seeding procedure shall be approved by the DRC prior to commencement of any seeding activities and such seeding shall be performed in strict compliance with the seeding procedure as specified by the DRC.
- 10.04 <u>Building Locations</u>. No building or other Structure shall be located on or built on any Lot nearer to the from Lot line or nearer to the side street right-of-way line than the minimum set back line. No building or Structure shall be placed nor shall any refuse or material, including but not limited to firewood, be placed or stored on any Lot within thirty (30) feet of the rear property line of any Lot.
- 10.05 <u>New Construction</u>. All Living Units and other Structures permitted hereby shall be new construction and no buildings shall be moved onto any Lot.

10.06 <u>Uncompleted Structures</u>.

- (a) Construction of a Living Unit or any other Structure shall not commence until the DRC has approved the final plans and specification for such Living Unit or any other Structure.
- (b) Construction of each residence shall proceed diligently and without delay or interruption until completion.

- (c) No Living Unit or other Structure shall be permitted to stand with its exterior in an unfinished condition for a period longer than twelve (12) months after commencement of construction.
- (d) Extensions for periods beyond twelve (12) months may be granted by the DRC in its sole discretion
- (e) In the event of fire, windstorm, or other damage, no Living Unit or other Structure shall be permitted to remain in a damaged condition for more than three (3) months.
- (f) No Living Unit or other Structure shall be occupied until completed according to the plans and specifications approved by the DRC.
- 10.07 <u>Structures</u>. No temporary building, trailer, tent, garage, barn or other building, whether in the course of construction or otherwise, shall be placed upon any Lot. No detached Structure for purely ornamental purposes nor above ground swimming pools nor any permanently constructed stoves, grills or ovens may be erected on any part of any Lot without the consent of the DRC.
- 10.08 <u>Fences</u>. No fences or walls shall be placed on any Lot without permission of the DRC and no approved fence or wall shall be erected or maintained in such a manner as to obstruct the view of vehicular traffic. No wire or chain link fence shall be erected on any Lot.
- 10.09 <u>Placement of Signs on Property</u>. No sign, billboard or other advertising device of any nature shall be placed upon any Lot, including property identification signs, except by the Developer and except as provided herein and as are approved by the DRC. The DRC may adopt and promulgate rules and regulations relating to signs which may be used within the Property. If approved by the DRC as to color, location, nature, content, size and other characteristics, "For Rent" and "For Sale" signs shall be permitted to be placed upon any Lot provided that such signs have first been approved by the DRC.
- 10.10 <u>Keeping of Animals.</u> No animals or birds, domesticated or wild, other than customary household pets, shall be kept or maintained on any Lot, except as specifically authorized by the Design Review Committee. Further, such animals shall not be kept or maintained on common grounds or ponds or on property adjacent to or bordering Eaglesgate property. Violation of any of these requirements may subject the owner to a fine of \$100 and related fees for each occurrence; failure to pay these fines and fees on demand or to properly and promptly address the animal problem cited may incur other penalties, including loss of neighborhood privileges and a lien upon property. The following lists violations of this Declaration. The DRC may from time to time publish and impose other regulations setting forth the type and number of animals that may be kept on any Lot.
 - (a) In no event, shall any pets be kept, bred, or maintained within Eaglesgate for any commercial purpose, commercial use, in such a manner as to constitute a nuisance, or cause unsanitary conditions.

- (b) No animal of any kind shall be kept, walked, or exercised on landscaped Common Property. Further, owners of animals are required to clean up after their pets on sidewalks, streets, landscaped common areas, or private property.
- (c) Dogs or other household pets with aggressive or vicious propensities, or restricted by local ordinances, or of an exotic type or breed, are specifically prohibited.
- (d) The DRC may prohibit any dogs outside any Lot except on a leash and accompanied by a responsible person. Other restraint requirements include the following:
 - (i) No dog shall be allowed on an owner's property unattended and shall be appropriately restrained or controlled by an appropriate device.
 - (ii) No dog shall be allowed to be off the owner's premises without being attended by an owner or a person designated by the owner. Such animals shall be restrained by a leash and collar or harness.
 - (iii) No GPS or other tracking system meets the restraint requirements.
- (e) No outside animal shelters of any type whatsoever shall be located within any Lot. No dog run, kennel, or fenced area shall be permitted on any Lot, or adjacent to or bordering a Lot.
- (f) No dog over six (6) months of age shall be kept by any Resident unless such animal shall have a rabies inoculation and a proper license.
- (g) The DRC shall have the right to prohibit the keeping of any domestic or wild animal on any Lot, which animal continually barks, howls, or make any other noises so as to, in the opinion of the DRC, unreasonably disturb the peace of any Resident.
- 10.11 <u>Disposition of Trash and Other Debris.</u> No Lot shall be used or maintained as a dumping ground for rubbish. No lumber, metals, bulk materials, refuse or trash shall be kept, stored or allowed to accumulate on any Lot, except building materials during the course of construction for a period not to exceed one hundred eighty (180) days (commencing from day one of the first delivery of any of such materials) unless extended by the DRC in its sole discretion, for any approved Structure, unless such materials are screened from view in a manner approved by the DRC. During the course of construction, it shall be the responsibility of each Owner to insure that construction sites are kept free of unsightly accumulations of rubbish and scrap materials, and that construction materials, trailers, shacks and the like are kept in a neat and orderly manner. No burning of any trash, leaves, grass or weeds and no accumulation or storage of litter of any kind shall be permitted on any Lot. If trash or other refuse is to be disposed of by being picked up and

carried away on a regular and recurring basis, containers may be placed in the open, on any day that a pickup is to be made, as such place on the Lot as to provide access to persons making such pickup. At all other times, such containers shall be stored in such a manner that they cannot be seen from adjacent and surrounding property. All such containers shall be kept in a clean and sanitary condition. The DRC may adopt and promulgate reasonable rules and regulations relating to the size, shape, color and type of containers permitted and the manner of storage of the same on the Property.

- 10.12 <u>Parking of Motor Vehicles, Boats, and Trailers</u>. For reasons of safety, traffic flow, and the appearance of the neighborhood, residents of Eaglesgate, their guests, and others shall observe these requirements regarding the parking and storage of vehicles in the neighborhood. In the event of a violation of this section, the Association Board or DRC shall give written notice to the Owner of the Lot that a vehicle in violation must be removed within 24 hours. If the vehicle is not removed within that time, the Association Board or DRC shall impose an Individual Assessment in accordance with Section 15.02 or other penalties against the Owner of the Lot.
 - (a) No truck, pickup, van, commercial vehicle or trailer, recreational vehicle, all-terrain vehicle, camper, motorcycle, automobile, mobile home, boat, or boat trailer shall be brought upon, stored or habitually parked on any Lot in front of any Living Unit or garage, or between any residence or garage and an abutting side street, or upon any street abutting any Lot. This shall not be construed to prohibit the mere temporary (a maximum of twenty-four (24) hours):
 - (i) standing or parking of a trailer, boat, trailer house, recreation vehicle, or mobile home for short periods preparatory to taking the same to some other location for use; or
 - (ii) standing or parking of a truck or commercial vehicle for loading, or unloading; or
 - (iii) parking of any operational automobile on any driveway on any Lot.
 - (b) The Association, with the written approval of the DRC, may permit operational vehicles to park on the street or in a driveway temporarily for loading and unloading or for preparation for use. In addition, trailers of any kind (boat, utility, travel, etc), boats, RVs, campers, and commercial vehicles also may be parked in a driveway, or in front of a house, for loading and unloading purposes and while service or construction is being performed. The 24-hour time limit applies.
 - (c) Motorcycles, ATVs, off-road vehicles, three-wheel vehicles, four-wheel vehicles (such as golf carts), Segways, and similar vehicles may not be operated on Eaglesgate common grounds or sidewalks. When used on the streets, operators of such vehicles shall adhere to usual traffic rules. None of these and similar vehicles shall be stored in any driveway, street, or yard

- outside of the timeframe for their immediate use. None of these and similar vehicles shall be openly stored for longer periods in any area other than as may be designated by the DRC.
- (d) No mechanical maintenance or service is to be performed on any vehicle outside of the garage. No mechanical maintenance on any vehicle shall be permitted in front of any Living Unit or garage, or between any Living Unit or garage and an abutting side street, or upon any street abutting any Lot.
- (e) Mobile homes may not be parked or stored in Eaglesgate.
- (f) Temporary storage containers, such as Portable On Demand Storage (PODS), dumpsters, and similar containers cannot be placed on driveways, streets in front of a Lot, or elsewhere on a lot without written permission of the DRC and not for more than fourteen (14) days.
- (g) Parking in the clubhouse parking lot is for users of Association facilities only. Others must obtain permission from the ECA Board of Directors.
- (h) Mobile office trailers may be parked on a lot during construction with prior DRC approval. While nothing contained herein shall be considered to prohibit the use of portable or temporary buildings or trailers as field offices by contractors during actual construction in Eaglesgate, the use and appearance of such a building or trailer must be specifically approved by the DRC prior to its being moved on site.
- (i) For purposes of this Section, the following definitions apply:
 - (i) All-terrain vehicle (ATV): An off-road vehicle that is not intended, or legal, for street use. Often referred to as a quad runner or a side-by-side.
 - (ii) Automobile: An operational vehicle that does not fall into a category listed above. Often referred to as a daily driver.
 - (iii) Recreational Vehicle (RV): A vehicle that includes a bathroom, kitchen, and beds that is intended for camping.
 - (iv) Camper: A towed vehicle with sleeping and cooking facilities, intended for camping.
 - (v) Mobile Home: A dwelling on a steel chassis with wheels, intended for permanent placement.
 - (vii) Commercial Vehicle: Any vehicle greater than 3/4 ton; any truck with an enclosed box, such as a delivery truck; any vehicle with

- affixed tool boxes or tool racks; any vehicle with commercial lettering or signage beyond the rear glass
- (viii) Habitually parked: Parking on the street by residents or their guests in the same or similar areas for more than 24 non-continuous hours without DRC approval.
- 10.13 <u>Nuisances</u>. No noxious or offensive activity shall be carried on upon any portion of the Lots, Living Units, or common grounds, nor shall anything be done thereon that may be or become a nuisance or annoyance to any other Owners.
 - (a) No exterior lighting shall be directed outside the boundaries of any Lot but shall be directed so as to avoid glare and excessive light spillage onto abutting or adjacent Property or Lots. Exterior lighting shall consist of concealed sources of illumination and shall maintain lighting levels consistent with the recognized standards of the lighting industry. Exterior lighting shall be from white sources only.
 - (b) Upon notice from the DRC that an exterior light is objectionable, an Owner shall immediately shield such light in such a manner so that in the opinion of the DRC the light is no longer objectionable. If shielding cannot be accomplished to the satisfaction of the DRC or the light continues to be objectionable, the DRC may require that such light be removed or replaced with a light that is not objectionable.
 - (c) Notwithstanding the foregoing, temporary, decorative lighting shall be permitted, provided that such lighting conforms to the requirements and limitations as may be imposed by the Association.
 - (d) Excessive noise, including amplified music, television programming, dogs barking, outside activities, disturbance of the peace, and other sounds unduly intruding on residents, especially after hours, shall be avoided.
 - (e) Residents shall keep pets under control and clean up after them on sidewalks, common grounds, and private property.
 - (f) Remote control devices, such as unmanned aerial vehicles (drones), that are capable of injuring people or pets, violating property lines, invading privacy, damaging property, or are unduly loud shall not be operated in an intrusive or unsafe manner.
- Antennas, Poles, and Projections. No Facilities, including poles and wires for the transmission of electricity, telephone messages, CATV signals, and the like, shall be placed or maintained above the surface of the ground on any Lot, and no external or outside antennas shall be permitted on any Lot or Living Unit without the prior written approval of the DRC. Satellite dishes are permitted, but the DRC may review placement of dishes on homes. No solar collectors of any

kind or type shall be installed or maintained. No flag poles, poles, or standards shall be erected or maintained, except with the prior written permission of the DRC. Requests for placement of basketball goals at homes must be approved by the DRC, which will issue requirements for location, type, color, and maintenance.

- 10.15 <u>Surface Water</u>. No well, pump, shaft, casing or other facilities for the removal of subsurface water shall be placed or maintained on any Lot, nor shall any boring, drilling, removal of or exploration for subsurface water be conducted on any Lot, except by or with the permission of the DRC. No individual water supply system shall be permitted on any Lot.
- 10.16 <u>Drainage</u>. Drainage from a Living Unit or Lot directly on to an adjoining Lot as a result of any construction activity or any change to the grade of any Lot shall be prohibited and each Owner shall be required to maintain the Lot and to construct and maintain the gutters and downspouts to control such drainage. The final grading on each Lot shall not cause any adverse change (as determined solely by the DRC) to the natural grade of such Lot.
- 10.17 <u>Sanitary Sewers</u>. No individual sewage treatment system shall be permitted on any Lot. All sanitary sewer lines shall connect with the central sewage disposal system provided. Water from downspouts or any surface water shall not be permitted to drain into the sanitary sewer system.
- 10.18 <u>Air and Water Pollution</u>. No use of any Lot (other than the normal use of Living Unit fireplaces and chimneys) will be permitted which emits pollutants into the atmosphere, or discharges liquid or solid wastes or other harmful matter into any waterway in excess of environmental standards applicable thereto, to be established by the DRC, which standards shall at a minimum meet the requirements of federal and state law and any regulations thereunder applicable to the Property. No waste or any substance or materials of any kind shall be discharged into any private or public sewer serving the Property, or any part thereof, in violation of any regulations of the State of Missouri or any private or public body having jurisdiction. The burning of leaves, trash or any debris is specifically prohibited.
- Mining and Drilling. No Lot or portion thereof shall be used for any mining, boring, quarrying, drilling, removal of, or any other exploitation of subsurface natural resources, with the sole exception of subsurface water, except for areas specifically designated for such purposes by the DRC. No oil drilling, oil development operation, oil refining, quarry or mining operations of any kind shall be permitted upon any Lot, nor shall oil wells, tanks, tunnels, mineral excavations or shafts be permitted upon any Lot. No derrick or other Structure designed for use in boring oil or natural gas shall be erected, maintained or permitted upon any Lot.
- 10.20 <u>Placement of Pipeline</u>. No water pipe, gas pipe or drainage pipe shall be installed or maintained on any Lot above the surface of the ground, other than as may be approved by the DRC, except at the point of connection of such pipe to Living Unit service and except for hoses used for the watering of lawns.

- 10.21 <u>Commercial Activity</u>. No commercial activity of any kind shall be conducted in any Residential Area or in any Living Unit, but nothing herein shall prohibit the carrying on of promotional activities by the Developer.
- 10.22 <u>Fireworks and Use of Firearms</u>. The sale and use of fireworks of any kind whatsoever on the Property is prohibited. Except as permitted by law for security personnel, the use of or discharge of firearms of any kind whatsoever is prohibited, except by permit granted to individuals by the Association for the purpose of target practice and trap or skeet shooting. The Association may set aside certain areas for this use if, in its sole discretion, it deems such action appropriate and desirable. Hunting of any kind, and by any method, including but not limited to firearms, traps, snares, bow and arrows, or manually propelled missiles is prohibited.
- 10.23 <u>Laws and Ordinances</u>. Each Owner shall promptly comply with all laws and statutes, ordinances, rules and regulations of federal, state or municipal governments or authorities applicable to use, occupancy, construction and maintenance of improvements upon any Lot or Living Unit.
- 10.24 "Off Road" Vehicular Traffic. None of the Property, including but not limited to the Common Property shall be used for motorized vehicular traffic of any nature except as to maintenance vehicles used in the ordinary course of maintaining the Property. Such prohibition extends to vehicles generally referred to and categorized as all terrain vehicles, motorcycles, motorized bikes and all other such motorized vehicles.
- 10.25 <u>Penalties for Violation of Article X</u>. If the DRC determines that provisions of this Article have been violated, the DRC may in its discretion seek appropriate relief at law or in equity to assure that the purposes of this Article are fulfilled, including those specified in Section 8.12, 14.01, 14.02, 14.03, and 14.04 hereof.

ARTICLE XI

Residential Protective Covenants and Restrictions

- 11.01 <u>Residential Provision</u>. The provisions of the Article XI shall relate to all Lots on the Eaglesgate Plat.
- 11.02 <u>Restrictions for Residential Lots</u>. No profession or home industry shall be conducted in or on any part of a Lot or in any improvement thereon in the Residential Areas without the specific written approval of the DRC. The DRC, in its discretion upon consideration of the circumstances in each case, and particularly in consideration of the effect on surrounding Property, may permit a Lot or any improvement thereon to be used in whole or in part for the conduct of a profession or home industry. No profession or home industry shall be permitted, however, unless, it is considered by the DRC to be compatible with the neighborhood. Except as provided herein, any Living Unit located on a Lot designated for residential use within the Property shall occupied by the Owner of the Living Unit. Notwithstanding the above, however, leasing of single family or

condominium units will be allowed provided that the Owner first register the lessee with the Association, providing such information as the Association may require, and file the Association a complete copy of the executed lease agreement.

- 11.03 <u>Lot Use for Model Home or Real Estate Office</u>. All else herein to the contrary notwithstanding and as expressly limited herein, any residential Lot may be used for a model home or for a real estate office by the Developer during the Development Period. Such right shall be limited to the Developer and shall extent to no other person, builder, Owner or other developer except as may be permitted by a majority vote of the Association Board.
- 11.04 <u>Use of Clothes Hanging Devices and Machinery</u>. No clothing or any household fabrics shall be hung in the open on any Lot. No machinery shall be operated upon any Lot (except such machinery used in the maintenance of a private residence) except with written approval of the DRC. No machinery shall be placed, parked or stored upon any lot unless such machinery is placed, parked or stored within an approved Structure. No hoisting devices shall be permitted upon any Lot; except, however, with the prior written approval of the Association, subject to review and written approval of the DRC.
- 11.05 <u>Provisions Applicable to Lots Designated for Single-Family Residences</u>. All Lots in Phases I, II, III, and IV on the Plat shall be subject to the following use restrictions, in addition to other provisions herein.
 - (a) <u>Land Use</u>. No Lots may be improved, used, or occupied for other than residential purposes and no flat or apartment house, although intended for residential purposes, may be erected or operated thereon.
 - (b) <u>Size Requirements</u>. To reflect replats, mergers, and other property changes, the Design Review Committee shall periodically update lot sizes and square footage requirements for new homes and list them in the Eaglesgate Development Guidelines. The DRC will permit less square footage exceptions to the requirements in the Guidelines in a consistent manner, and not on a spot basis, taking into consideration the size, shape, and use of adjoining property.
 - (c) <u>Building Lines</u>. No part of any residence shall be located on any Lot nearer to the front street or the side street than the front building line (setback) or the side building line, unless otherwise shown on any recorded plat, or as otherwise shown or permitted in the Development Guidelines, or as permitted by the DRC. Further, no part of any residence shall be located on any Lot nearer than six (6) feet to the side property lines. Each residence shall be placed on each Lot in compliance with all governmental requirements and ordinances.
 - (d) <u>Garages</u>. Each residence shall have an attached, private, fully enclosed garage for not less than two (2) nor more than four (4) cars. The driveway

constructed on each Lot shall contain sufficient paved area for the off- street parking of at least two (2) cars. Any garage facing a street shall be equipped with doors that shall be closed as much as practicable to preserve the appearance of the elevation of the residence fronting on the street. Garages may be used only for the storage of automobiles and related use as a storage facility and for no other purposes, i.e., no garage space shall be enclosed for use as living space.

- (e) Frontage. Each residence shall front and present a good frontage on the street on which it is located as shown on the Plat. Each residence located on a corner Lot shall front or present a good frontage on both streets. On corner Lots as designated by the DRC, the residence must be set at a forty-five (45) degree angle as determined by the DRC. Each residence shall present good frontage on all four (4) sides.
- (f) <u>Building Materials</u>. All roofs shall be made of materials in conformity with the Eaglesgate Development Guidelines. Roof pitches must be consistent with the architectural style of the building. Exterior building materials may be combinations of brick, stone, wood siding, wood paneling, stone, masonite horizontal lapped siding which is painted, plate glass or stucco, or a combination thereof. No masonite board or batt, or lava rock, exposed concreted block, simulated brick, or wood for exterior walls are allowed without the prior approval of the DRC. Building materials must remain consistent on all exterior elevations.

ARTICLE XII

Construction on Lots

12.01 Construction Standards.

- (a) Commencement of construction on a Lot shall start within sixty (60) days following the recording of the deed from the Owner to the purchaser. Construction shall proceed in a timely and orderly manner to a prompt completion.
- (b) No Lot is to be cleared nor shall construction commence on any Lot until a building permit therefore is granted, the Lot closing has taken place and the DRC has approved the plans and specifications for such construction.
- (c) No dumping or open burning of construction materials, waste or trash shall occur on any building Lot.
- (d) Loud music will not be permitted on any construction site.
- (e) No construction signs are permitted identifying the home builder, subcontractors or supplies.

- (f) Certain tree protection procedures have been formulated for Eaglesgate. It should be understood that compacting of soil, trenching and grade changes involving cutting or filling often causes death to a healthy tree within three to four years. The following procedures are, therefore, recommended;
 - (i) Tree clearing of building sites should be done by hand to minimize disturbance of remaining trees.
 - (ii) Grading or trenching within the dripline of trees should be minimized and preferably limited to areas away from the center of the tree crown. A qualified arborist or landscape architect should be consulted when working within the dripline of major (as determined by the DRC) trees.
 - (iii) A qualified arborist should also be consulted if overhead branches of major trees interfere with the construction of the dwelling.
 - (iv) A four-foot construction fence should be erected at the dripline of major trees and tree groupings. No construction activities including storage of materials or parking of vehicles or equipment should be allowed within the dripline of trees. Signs, bracing, and temporary wiring should not be nailed to any tree.
- (g) Erosion control shall be provided on Lots with steep grades. The DRC may, at its sole discretion, require the builder to place erosion control materials such as straw bales or fencing on any portion of a Lot that appears to be in an erodible condition due to construction activities.
- (h) Builders and contractors are responsible for the actions of their workers as well as those of their subcontractors.
- (i) No changes in plans during the construction period will be permitted without prior express written approval of the DRC.
- (j) Once residences are occupied within the Eaglesgate community, no construction work shall begin before 7:00 a.m. or continue after 7:00 p.m. Exterior construction on new homes is not permitted on Sundays. Interior work on new homes and normal maintenance on existing structures is permitted on Sundays.
- (k) Excess excavation materials must be hauled away from the Lot and from the Property.
- (l) Concrete suppliers and contractors shall clean their equipment only at locations designated by the Developer for that purpose.
- (m) Builders shall clean up all trash and debris on the construction site at the end of each week. Trash and debris shall be removed from each construction site at least once a week to a dumping site located off the Property. Builders and

their subcontractors will be responsible for removing all construction debris and keeping construction sites in a well-maintained appearance at all times.

ARTICLE XIII

Duration and Amendment

13.01 <u>Duration</u>. This Declaration and the Restrictions contained herein shall run with, burden and bind the Property, shallinure to the benefit of all, and shall be enforceable by the Association and any Owner, their respective legal representatives, heirs, successors and assigns, and by any Resident until August 1, 2013; after which time the Declaration shall be automatically renewed for successive periods of ten (10) years unless, prior to the commencement of any such renewal period, an instrument terminating this Declaration and the Restrictions contained herein shall be executed by the proper Association officers and recorded in the appropriate Office of the Recorder of Deeds or in such other place of recording as may be appropriate at the time of the execution of such instrument, pursuant to a resolution to such effect approved in writing by not less than two-thirds (2/3) of the total number of Members under the provisions of Section 3.07, which resolution shall have been approved within six (6) months prior to August 1, 2013, or the end of any such ten (10) year extension period.

13.02 Amendment.

- (a) Except as hereinafter specifically provided, this Declaration may not be amended, terminated, or modified in any respect except by recording an instrument executed by the proper Association officers and authorized by the Members, subject of course to the rights, if any, of any lienholders of liens on the Property to consent to or approve of such amendment, termination or modification, pursuant to a resolution to such effect approved in writing by not less than two-thirds (2/3) of the total number of Members under the provisions of Section 3.07.
- (b) Notwithstanding the foregoing, this Declaration can be abolished, modified, or changed in whole or in part by the Association in order to, among other things, correct deficiencies of this Declaration as determined to exist by the Association; to annex property as provided for herein or to de-annex Property without Association membership approval but with the written consent of Owners located within the boundaries of the Property to be deannexed (who together with the Association shall execute a release document for recording with the appropriate Office of the Recorder of Deeds) and, to give effect to all of the rights, obligations and duties created or contemplated herein.
- (c) Notwithstanding the foregoing, any amendment or modification of this Declaration and the restrictions contained herein which would change any obligation of the Association to maintain any Common Property, common

area drainage easement, retention pond, or any landscaping within the rightor-way of any public or private streets within the boundaries of the Property as depicted on the final plat shall require the written approval of Greene County, Missouri or the City of Springfield, Missouri if the Property is subsequently annexed into the City of Springfield, before it shall become effective. No amendment shall be effective until it is recorded in the Recorder of Deed's Office of Greene County, Missouri.

ARTICLE XIV

Enforcement

- Right of Action. In the event of a violation or breach of any Restriction contained in 14.01 this Declaration, the Association shall give written notice to the Owner setting forth in reasonable detail the nature of such violation or breach and the specific action or actions which shall be taken by the Owner to remedy such violation or breach. If the Owner shall fail to take reasonable steps to remedy such violation or breach within the time limit specified in the written notice, then the Association may pursue its Right of Action. The term "Right of Action" as used herein, shall mean the right of the Association, through its agents and employees, to enter at all reasonable times upon any Lot as to which a violation, breach or other condition to be remedied exists, and take the actions specified in the notice to the Owner to abate, extinguish, remove or repair such violation, breach or other condition which may exist thereon contrary to the provisions hereof. Such entry or action, or both, shall not be deemed to be a trespass or wrongful act solely by reason of such entry and such actions, provided such entry and such actions are carried out in accordance with the provisions of this Section. All costs and expenses including reasonable attorney's fees incurred by the Association or on its behalf in enforcing such Right of Action, shall be a binding personal obligation of such Owner enforceable at law, as well as a lien on such Owner's Lot enforceable pursuant to the provisions of Section 14.03 hereof. The lien provided under this Section shall not be valid against a bona fide purchaser (or bona fide lienholder) of the Lot in question unless a notice of such lien shall have been filed in the appropriate office of the Recorder of Deeds prior to the recordation of the Deed in the said office (or lien instrument) conveying the Lot in question to such purchaser (or subjecting the same to such lien). "Right of Action" shall also mean and encompass the right to pursue all remedies herein specified and specified in Sections 14.02 and 14.03, together with all remedies at law or in equity.
- 14.02 <u>Individual Assessments</u>. The Board shall have the authority to levy from time to time an Individual Assessment against any Lot and its Owner to correct or eliminate any breach by such Owner of any agreement, obligation, reservation, or restriction contained in any deed, declaration, or plat covering such Lot (including without limitation, to maintain or repair any Lot or improvement thereon), the amount of such assessment to be equal to the greater of
 - (1) any amount expended by the Eaglesgate Community Association to correct or eliminate such breach, or

(2) \$25 per day that such breach continues following ten days written notice of such breach to the Owner (unless a different notice period is specified herein with regard to such breach, in which case such different notice shall apply.

Each such Individual Assessment shall be due and payable upon giving notice of the assessment to such Owner.

14.03 Specific Performance. Nothing contained herein shall be deemed to affect or limit the rights of the Association, the Members, the Residents or the Owners, or any one of them, to enforce any of the terms, covenants or conditions of the Declaration by appropriate judicial proceedings. However, the Association hereby declares that it is impossible to measure in money the damages which will accrue to a beneficiary hereof, its transferees, successors or assigns, by reason of a violation of, or failure to perform any of the obligations provided by this Declaration. Therefore, any beneficiary hereof shall be entitled to relief by way of injunction or specific performance, as well as any other relief available at law or in equity, to enforce the provision hereof. In any and all such actions, whether at law or in equity, any such beneficiary hereof who is entitled to relief shall also be entitled to recover all costs and expenses, including reasonable attorney's fees, incurred in enforcing such rights.

14.04 Enforcement of Liens.

- (a) The Association shall have a lien for Assessments, user fees, and charges (herein collectively, "Assessment" or "Assessments") as set forth in Section 4.01 hereof and shall have a lien for the cost of exercising the Right of Action as set forth in Section 14.01 hereof. The amount which may be recovered by the Association shall include the Assessment or costs, together with the cost of such enforcement proceedings, including reasonable attorney's fees and interest. Suits to recover a money judgment for unpaid Assessments shall be maintainable without foreclosing or waiving the lien provided for in Section 4.01 hereof.
- (b) If any demand for payment or claim of lien or liens is not paid when due as provided in Section 4.06 hereof, the Association Board or its duly authorized representative may thereafter elect to file and record a claim of lien on behalf of the Association against the Lot or Living Unit of the defaulting Owner in the appropriate Office of the Recorder of Deeds. Such claim of lien shall be executed and acknowledged by any officer of the Association and shall contain substantially the following information:
 - (i) the name of the delinquent Owner;
 - (ii) the legal description and street address of the Lot or Living Unit against which the claim of lien is made;

- (iii) the total amount claimed to be due and owing for the amount of the delinquency, interest thereon, collection costs and reasonable attorney's fees (with any proper offset allowed);
- (iv) that the claim of lien is made by the Association pursuant to this Declaration; and
- (v) that a lien is claimed against said Lot or Living Unit in an amount equal to the amount stated; together with all other amounts becoming due from time to time in accordance with this Declaration.
- (c) Upon such recordation of the duly executed original or copy of such a claim of lien, the lien claimed therein shall immediately attach and become effective in favor of the Association as a lien upon the Lot or Living Unit against which such Assessment or cost was levied. Such a lien shall have priority over all liens or claims created subsequent to the recordation of the claim of lien thereof, except for tax liens for real property taxes and assessments on any Lot or Living Unit in favor of any municipal or other governmental unit and except as provided in Section 4.01 hereof.
- (d) Any such lien may be foreclosed by appropriate action at law or equitable remedies. The Association Board is hereby authorized to appoint any attorney or any officer or director of the Association for the purpose of conducting such proceeding.
- (e) The lien provided for herein shall be in favor of the Association and shall be for the benefit of all other Owners and shall secure payment of all sums set forth in the claim of lien, together with all sums becoming due and payable in accordance with this Declaration after the date of recordation of said claim of lien.
- (f) Upon the timely curing of any default for which a notice of claim of lien was filed by the Association Board and the payment of all sums secured by the lien created by the recordation of such claim of lien, the Association Board shall (upon payment by such Owner of reasonable costs by the Owner of the Lot or Living Unit subject to the lien) cause an officer of the Association to file and record an appropriate release of such claim of lien in the appropriate Office of the Recorder of Deeds.
- (g) No Owner may waive or otherwise escape liability for the Assessments provided for in this Declaration by nonuse of the Common Area, or any part thereof, or any part of the Property, or abandonment of his Lot or Living Unit. Notwithstanding anything contained in this Declaration to the contrary, no action may be brought to foreclose or otherwise realize on the lien created by recordation of the claim of lien, until the expiration of thirty (30) days after a copy of said claim of lien, showing the date of recordation thereof,

- has been mailed to the Owner of the Lot or Living Unit which is described in such claim of lien.
- (h) Each Owner does hereby waive to the extent legally possible, all defenses to any liens created pursuant to this Declaration, whether such liens are now in existence or are created any time in the future, and the benefit of any exemption laws of the State of Missouri now in effect, or in effect from time to time hereafter.
- 14.05 <u>No Waiver</u>. The failure of the Association, any Owner, his or its respective legal representatives, heirs, successors and assigns, or any Resident, to enforce this Declaration shall in no event be considered a waiver of the right to do so thereafter as to a similar violation of breach occurring prior or subsequent thereto.
- Additional Rules. The Association Board, and the DRC, each by a majority vote, to the extent specifically provided herein, may adopt, amend, modify, promulgate and rescind or revoke reasonable rules, regulations and procedures regarding the administration, interpretation and enforcement of the provisions of this Declaration; for the DRC in particular, this includes the right of modification, amendment, revocation or recision of the Development Guidelines. In so adopting, amending, modifying, promulgating, rescinding or revoking such rules, regulation and procedures, or in making any finding, determination, ruling or order or in carrying out any directive contained herein relating to the issuance of permits, authorizations, approvals, rules or regulations, the Association and the DRC shall take into consideration the best interests of the Owners and Residents of the Property to the end that the Property shall be preserved and maintained as a community of high quality, and shall seek to achieve the development of the Property in accordance with the standards and objectives set forth in the Development Plan.

14.07 Incorporation of Provisions in Deeds.

- (a) Each grantee, by accepting a Deed, lease or other instrument conveying any interest in any Lot, whether or not such instrument incorporates or refers to this Declaration, covenants for himself or itself, its heirs, successors and assigns to observe, perform and be bound by the Declaration and to incorporate this Declaration by reference in any Deed or other conveyance of all or any portion of his interest in any real property subject hereto.
- (b) Each grantee taking title by acceptance of a Deed, lease or other instrument conveying any interest in any Lot further agrees to cause all subsequent grantees to execute any Deed, lease or other instrument conveying any interest in any Lot for the purpose of affirmatively assuming the obligations of an Owner hereunder and agrees to include the following covenant in any such Deed or other instrument conveying any interest in any Lot:

For the benefit of the grantor, the Eaglesgate Community Association, and its respective heirs, successors, and assigns, the grantee hereunder executes this instrument for the purpose of assuming

the obligations of an Owner under the Declaration of Covenants, Restrictions, Easements, Charges and Liens to which the property is subject and expressly agrees to comply with each provision thereof to the extent such provision applies to him or it.

This covenant, and any such covenant in any deed to any Lot, may be specifically enforced against the granter or the grantee, or both.

ARTICLE XV

Miscellaneous

- 15.01 <u>No Reverter</u>. No Restriction herein is intended to be, or shall be construed as, a condition subsequent or as creating a possibility of reverter.
- 15.02 <u>Invalidity</u>. The determination by a court that any provision hereof is invalid for any reason shall not affect the validity of any other provision hereof, and to the extent that any term, covenant or condition contained in this Declaration is in conflict with any applicable laws, this Declaration shall be deemed to be amended so as to comply with applicable laws.
- 15.03 <u>Violation and Nuisance</u>. Any act or omission whereby any provision of this Declaration is violated in whole or in part is hereby declared to be a nuisance and may be enjoined or abated, whether or not the relief sought is for negative or affirmative action, by the Developer, the Association or any Owner of a Living Unit or Lot.
- 15.04 <u>Violation of Law</u>. Any violation of any federal, state, municipal or local law, ordinance or regulation pertaining to the ownership, occupation or use of any of the Property is hereby declared to be a violation of this Declaration and subject to any and all of the enforcement procedures set forth herein.
- 15.05 <u>Remedies Cumulative</u>. Each remedy set forth in this Declaration shall be in addition to all remedies whether available at law or in equity and all such remedies, whether or not set for the in this Declaration, shall be cumulative and not exclusive.
- 15.06 <u>Limitations</u>. During the Development Period, the Association may not use its resources nor take a public position in opposition to the general Development Plan or to changes thereto proposed by the Developer. Nothing in this Section shall be construed to limit the rights of Members acting as individuals or in an affiliation with other members of groups.
- 15.07 <u>No Personal Liability</u>. No member of the Association Board, officer of the Association, member of the DRC, member of any committee of the Association, whether such committee is specifically described in this declaration or hereafter created by the Association, or Manager, if any, or the Developer shall be personally liable to any Owner, Member, Resident or to any other party for any damage, loss or prejudice suffered or claimed on account of any act, (including any oral representation regarding any aspect of a Lot whatsoever), omission, error, failure to act, or negligence of any such Association Board Member, officer or committee member

of the Association, Manager, if any, the Developer, or any Member of the DRC or any realtor representing the Developer in the sale of a Lot and, further, neither the DRC nor any member thereof shall be liable to the Association, any Owner or to any other party for any damage, loss or prejudice suffered by or claimed on account of (a) the approval or disapproval of any plans, drawing or specifications, whether or not defective, (b) the construction or performance of any work upon the Property, (c) the execution and filing of any estoppel certificate, whether or not the facts therein are correct, or (d) any other act, action or conduct of such committee. Such limitation of liability shall apply in all cases, provided that such person has, on the basis of such information as may be possessed by him, acted in good faith, without willful or intentional misconduct.

15.08 <u>View Lots</u>. The aesthetically appealing vista of certain Lots, as of the approximate date hereof, due to the location of such Lots or the absence of completed development within the Development Plan may be altered as a result of further development of the Property.

15.09 <u>Assignability</u>.

- (a) The Association shall be empowered to assign its rights, or any part thereof, to any successor public body, authority, agency, district or not-for-profit corporation (hereinafter referred to as the "Successor Entity"), and upon such assignment the Successor Entity shall have those rights and be subject to those duties assigned thereby and shall be deemed to have agreed to be bound by the appropriate provisions hereof to the same extent as if the Successor Entity had been an original party to the Declaration. Any such assignment shall be accepted by the Successor Entity under a written agreement pursuant to which the Successor Entity expressly assumes the duties and obligations thereby assigned.
- (b) If for any reason the Association shall cease to exist without having first assigned its rights hereunder to a Successor Entity, the covenants, restrictions, easements, charges and liens imposed hereunder shall nevertheless continue and any Owner or Resident may petition a court of competent jurisdiction to have a trustee appointed for the purpose of organizing a not-for-profit corporation and assigning the rights hereunder with the same force and effect, and subject to the same conditions, as provided in this Section 16.09 with respect to an assignment and delegation to a Successor Entity.
- (c) Any assignment or delegation of rights shall be approved by two-thirds (2/3) of the Members voting in person or by proxy at an Association meeting at which a quorum is present or voting in a referendum called for such purpose after proper notice is given.
- (d) The Association may, at its option, assign any or all or its rights under this Declaration.

- 15.10 <u>Headings</u>. The headings of the Articles and Section hereof are for convenience only and shall not affect the meaning or interpretation of the contents of this Declaration.
- 15.11 <u>Gender</u>. Throughout this Declaration, the masculine gender shall be deemed to include the feminine and neuter, and the singular, the plural, and vice versa.
- 15.12 <u>Effect of Violation of Declaration on Mortgage</u>. No violation of this Declaration shall defeat or render invalid the lien of any mortgage made in good faith and for value upon any portion of the Property; provided, however, that any mortgagee in possession or any purchase at any foreclosure sale or any person in a similar position shall be bound and subject to this Declaration as fully as any other Owner of any portion of the Property.

15.13 Delivery of Notices and Documents.

- (a) Any written notice or other documents addressed to the Association, the Association Board, the DRC, or the Developer relating to or required or permitted by the Declaration may be delivered either personally or by certified or registered mail, return receipt requested. If by certified or registered mail, it shall be deemed to have been given, delivered and received upon receipt thereof by the addressee.
- Any written notice or other documents relating to or required or permitted by (b) the Declaration may be delivered to an Owner or Member either personally or by mail unless other requirements are specifically made in any provision hereof. If by mail, it shall be deemed to have been given, delivered and received seventy-two (72) hours after a copy of same has been deposited in the United States mail, postage prepaid, addressed to such Owner or Member, to the address of any Lot or Living Unit owned, whether in whole or in part, by such Owner or Member, or to any other address last furnished by such Owner or Member to the Association. Each Owner or Member shall file his correct mailing address with the Association, and shall promptly notify the Association in writing of any subsequent change of address. Any notice given or required to be sent pursuant to this Section 16.13(b) shall be deemed to have been properly given, unless other requirements are specifically made in any provision hereof, when mailed, postage prepaid, to the last known address of the person to whom notice is to be given.
- 15.14 <u>Local Laws Not Superseded</u>. This Declaration shall not be construed as permitting any action prohibited by applicable zoning laws, or by the laws, rules or regulations of any governmental body, or by any deed or lease.
- 15.15 <u>Severability</u>. In the event any one or more of the provisions contained in this First Amended Declaration shall for any reason be held to be invalid, illegal, or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision, but each shall be construed as if such invalid, illegal, or unenforceable provision had never been included.

IN WITNESS WHEREOF, the following persons, who constitute the Board of Directors of the Eaglesgate Community Association, have executed this document on the 16-27 day of April, 2018. Jeff Mills				
Notary's Acknowledgement				
State of Missouri)) ss				
County of Greene)				
On this 26 day of April, 2018, before me personally appeared Jeff Mills, to me known to be the person described in and who executed the foregoing instrument and acknowledged to me that Jeff Mills executed the same as his free act and deed.				
My Commission Expires: //-3-20/8 BONNIE J. POWERS Notary Public - Notary Seal STATE OF MISSOURI Christian County Commission Number 14436787 My commission expires November 03, 2018 Notary's Acknowledgement				
1 (Ottaly 5 Memily Wedgement				
State of Missouri) ss County of Greene)				
On this 22 day of April, 2018, before me personally appeared Paul McMasters, to me known to be the person described in and who executed the foregoing instrument and acknowledged to me that Paul McMasters executed the same as his free act and deed.				
My Commission Expires: //-3-2018				
BONNIE J. POWERS Notary Public – Notary Seal STATE OF MISSOURI Christian County Commission Number 14436787 My commission expires November 03, 2018				

Brett Marti

Notary's	Acknowledgemen	nt
----------	----------------	----

State of Missouri)) ss			
County of Greene)			
On this day of April, 2018, before me personally appeared Brett Marti, to me known to be the person described in and who executed the foregoing instrument and acknowledged to me that Brett Marti executed the same as his free act and deed.			
My Commission Expires: /1-3-20/8			
BONNIE J. POWERS Notary Public – Notary Seal STATE OF MISSOURI Christian County Commission Number 14436787 My commission expires November 03, 2018 Susan Robinson			
Notary's Acknowledgement			
State of Missouri) ss			
County of Greene)			
On this 26 day of April, 2018, before me personally appeared Susan Robinson, to me known to be the person described in and who executed the foregoing instrument and acknowledged to me that Susan Robinson executed the same as her free act and deed.			
My Commission Expires: //-3 ·20/8			

BONNIE J. POWERS

Notary Public – Notary Seal
STATE OF MISSOURI
Christian County
Commission Number 14436787
My commission expires November 03, 2018

Notary Public Bonnie J. Powers

Arlyn Pémber

Notary's Acknowledgement

State of Missouri)	
)	SS
County of Greene)	

On this 272 day of April, 2018, before me personally appeared Arlyn Pember, to me known to be the person described in and who executed the foregoing instrument and acknowledged to me that Arlyn Pember executed the same as his free act and deed.

My Commission Expires: //-3-20/8

BONNIE J. POWERS

Notary Public – Notary Seal STATE OF MISSOURI Christian County Commission Number 14436787 My commission expires November 03, 2018 Notary Public Bonnie J. Powers

EXHIBIT "A"

Property Description Eaglesgate Phase 1

The property as fully described therein located on Book 2003, Page 082820-02, filed on November 17, 2003 with the Recorder of Deeds of County of Greene, State of Missouri.

Property Description Eaglesgate Phase 2

The property as fully described therein located on Book 2005, Page 028643-05, filed on June 1, 2005 with the Recorder of Deeds of County of Greene, State of Missouri.

Property Description Eaglesgate Phase 3

The property as fully described therein located on Book 2004, Page 070781-04, filed on December 29, 2004 with the Recorder of Deeds of County of Greene, State of Missouri.

Property Description Eaglesgate Phase 4

The property as fully described therein located on Book 2012, Page 023111-12, filed on May 31, 2012 with the Recorder of Deeds of County of Greene, State of Missouri.

EXHIBIT "B"

Property Description of Common Areas of Eaglesgate

A tract of land, situated in the Northwest Fractional Quarter of Section 30, Township 28 North, Range 21 West, Greene County, Missouri, and more particularly described as follows: Commencing at the Southwest Corner of said Northwest Fractional Quarter of Section 30; THENCE South 89 degrees 49 minutes 28 seconds East a distance of 51.00 feet to the point of beginning; THENCE North 00 degrees 27 minutes 32 seconds East a distance of 759.65 feet to a point for corner; THENCE North 45 degrees 27 minutes 27 seconds East a distance of 42.43 feet to a point for corner; THENCE South 00 degrees 27 minutes 32 seconds West a distance of 109.12 feet to a point for corner; THENCE South 59 degrees 08 minutes 02 seconds East a distance of 78.99 feet to a point for corner; THENCE South 73 degrees 09 minutes 35 seconds East a distance of 76.90 feet to a point for corner; THENCE North 75 degrees 15 minutes 54 seconds East a distance of 86.32 feet to a point for corner; THENCE North 46 degrees 37 minutes 02 seconds East a distance of 61.99 feet to a point for corner; THENCE North 42 degrees 08 minutes 49 seconds East a distance of 84.06 feet to a point for corner; THENCE South 63 degrees 59 minutes 52 seconds East a distance of 72.04 feet to a point for corner; THENCE South 42 degrees 08 minutes 49 seconds West a distance of 155.01 feet to a point for corner; THENCE South 06 degrees 10 minutes 36 seconds West a distance of 47.34 feet to a point for corner; THENCE South 48 degrees 31 minutes 59 seconds West a distance of 181.48 feet to a point for corner; THENCE South 07 degrees 07 minutes 57 seconds East a distance of 69.17 feet to a point for corner; THENCE South 78 degrees 41 minutes 05 seconds West a distance of 81.93 feet to a point for corner; THENCE South 35 degrees 43 minutes 09 seconds West a distance of 85.08 feet to a point for corner; THENCE South 22 degrees 47 minutes 31 seconds West a distance of 49.55 feet to a point for corner; THENCE South 11 degrees 34 minutes 02 seconds West a distance of 73.33 feet to a point for corner; THENCE South 00 degrees 12 minutes 06 seconds West a distance of 51.53 feet to a point for corner; THENCE South 00 degrees 12 minutes 19 seconds West a distance of 107.81 feet to a point for corner; THENCE North 89 degrees 49 minutes 28 seconds West a distance of 25.57 feet to the POINT OF BEGINNING, and containing 2.29 acres of land, more or less.

AND

A tract of land, situated in the Northwest Fractional Quarter of Section 30, Township 28 North, Range 21 West, Greene County, Missouri, and more particularly described as follows: Commencing at the Southwest Corner of said Northwest Fractional Quarter of Section 30; THENCE South 89 degrees 49 minutes 28 seconds East a distance of 51.00 feet; THENCE North 00 degrees 27 minutes 32 seconds East a distance of 884.65 feet to the point of beginning; THENCE continuing North 00 degrees 27 minutes 32 seconds East a distance of 361.12 feet to a point for corner; THENCE North 42 degrees 29 minutes 22 seconds East a distance of 33.07 feet to a point on a curve; THENCE along a curve to the left having a radius of 302.50 feet a delta of 21 degrees 27 minutes 21 seconds, an arc length 113.28 feet and a chord which bears North 72 degrees 22 minutes 19 seconds East having a chord distance of 112.62 feet to a point on a line;

THENCE South 79 degrees 34 minutes 53 seconds East a distance of 39.73 feet to a point on a curve; THENCE along a curve to the left having a radius of 195.00 feet a delta of 05 degrees 47 minutes 05 seconds, an arc length 19.69 feet and a chord which bears South 42 degrees 07 minutes 29 seconds East having a chord distance of 19.68 feet to a point of tangency; THENCE South 45 degrees 01 minutes 01 seconds East a distance of 137.01 feet to a point of curve; THENCE along a curve to the right having a radius of 95.00 feet a delta of 94 degrees 42 minutes 41 seconds, an arc length 157.04 feet and a chord which bears South 02 degrees 20 minutes 19 seconds West having a chord distance of 139.76 feet to a point of tangency; THENCE South 49 degrees 41 minutes 39 seconds West a distance of 54.03 feet to a point of curve; THENCE along a curve to the left having a radius of 125.00 feet a delta of 35 degrees 28 minutes 13 seconds, an arc length 77.38 feet and a chord which bears South 31 degrees 57 minutes 33 seconds West having a chord distance of 76.15 feet to a point on a line; THENCE South 14 degrees 12 minutes 50 seconds West a distance of 71.08 feet to a point for corner; THENCE South 52 degrees 20 minutes 15 seconds West a distance of 39.34 feet to a point for corner; THENCE North 89 degrees 32 minutes 28 seconds West a distance of 116.18 feet to a point for corner; THENCE North 44 degrees 32 minutes 23 seconds West a distance of 42.43 feet to the POINT OF BEGINNING, and containing 2.36 acres of land, more or less.

AND

A tract of land, situated in the Northwest Fractional Quarter of Section 30, Township 28 North, Range 21 West, Greene County, Missouri, and more particularly described as follows: Commencing at the Southwest Corner of said Northwest Fractional Quarter of Section 30; THENCE South 89 degrees 49 minutes 28 seconds East a distance of 51.00 feet; THENCE North 00 degrees 27 minutes 32 seconds East a distance of 1362.10 feet to the point of beginning; THENCE continuing North 00 degrees 27 minutes 32 seconds East a distance of 557.75 feet to a point on a curve; THENCE along a curve to the right having a radius of 1,450.00 feet a delta of 05 degrees 31 minutes 30 seconds, an arc length 139.82 feet and a chord which bears North 07 degrees 45 minutes 29 seconds East having a chord distance of 139.77 feet to a point of reverse curve: THENCE along a curve to the left having a radius of 1,550.00 feet a delta of 03 degrees 33 minutes 06 seconds, an arc length 96.08 feet and a chord which bears North 08 degrees 44 minutes 41 seconds East having a chord distance of 96.06 feet to a point on a line; THENCE South 89 degrees 27 minutes 09 seconds East a distance of 4.39 feet to a point for corner; THENCE South 01 degrees 07 minutes 03 seconds East a distance of 109.05 feet to a point on a curve; THENCE along a curve to the left having a radius of 50.00 feet a delta of 98 degrees 57 minutes 16 seconds, an arc length 86.35 feet and a chord which bears South 06 degrees 01 minutes 53 seconds West having a chord distance of 76.01 feet to a point of reverse curve; THENCE along a curve to the right having a radius of 15.00 feet a delta of 52 degrees 01 minutes 12 seconds, an arc length 13.62 feet and a chord which bears South 17 degrees 26 minutes 09 seconds East having a chord distance of 13.16 feet to a point of tangency; THENCE South 08 degrees 34 minutes 27 seconds West a distance of 52.45 feet to a point of curve; THENCE along a curve to the left having a radius of 295.00 feet a delta of 21 degrees 44 minutes 51 seconds, an arc length 111.97 feet and a chord which bears South 02 degrees 17 minutes 58 seconds East having a chord distance of 111.30 feet to a point of tangency; THENCE South 13 degrees 10 minutes 24 seconds East a distance of 34.54 feet to a point of curve;

THENCE along a curve to the right having a radius of 245.00 feet a delta of 27 degrees 35 minutes 28 seconds, an arc length 117.98 feet and a chord which bears South 00 degrees 37 minutes 20 seconds West having a chord distance of 116.84 feet to a point of reverse curve; THENCE along a curve to the left having a radius of 295.00 feet a delta of 52 degrees 18 minutes 47 seconds, an arc length 269.35 feet and a chord which bears South 11 degrees 44 minutes 20 seconds East having a chord distance of 260.09 feet to a point on a line; THENCE South 11 degrees 29 minutes 47 seconds West a distance of 30.82 feet to a point on a curve; THENCE along a curve to the right having a radius of 237.50 feet a delta of 15 degrees 49 minutes 58 seconds, an arc length 65.63 feet and a chord which bears South 71 degrees 34 minutes 41 seconds West having a chord distance of 65.42 feet to a point on a line; THENCE North 48 degrees 26 minutes 25 seconds West a distance of 37.92 feet to the POINT OF BEGINNING, and containing 0.69 acres of land, more or less.

AND

A tract of land, situated in the Northwest Fractional Quarter of Section 30, Township 28 North, Range 21 West, Greene County, Missouri, and more particularly described as follows: Commencing at the Southwest Corner of said Northwest Fractional Quarter of Section 30; THENCE South 89 degrees 49 minutes 28 seconds East a distance of 51.00 feet; THENCE North 00 degrees 27 minutes 32 seconds East a distance of 2153.76 feet; THENCE South 89 degrees 27 minutes 09 seconds East a distance of 139.10 feet to the point of beginning; THENCE continuing South 89 degrees 27 minutes 09 seconds East a distance of 116.61 feet to a point for corner; THENCE South 05 degrees 15 minutes 31 seconds East a distance of 117.35 feet to a point for corner; THENCE South 48 degrees 12 minutes 25 seconds East a distance of 210.24 feet to a point for corner; THENCE South 20 degrees 51 minutes 57 seconds East a distance of 118.29 feet to a point on a curve; THENCE along a curve to the left having a radius of 355.00 feet a delta of 46 degrees 33 minutes 07 seconds, an arc length 288.43 feet and a chord which bears South 47 degrees 33 minutes 47 seconds West having a chord distance of 280.56 feet to a point of tangency; THENCE South 24 degrees 17 minutes 14 seconds West a distance of 99.91 feet to a point for corner; THENCE North 23 degrees 43 minutes 48 seconds West a distance of 56.49 feet to a point for corner; THENCE North 00 degrees 24 minutes 16 seconds East a distance of 95.11 feet to a point for corner; THENCE North 00 degrees 24 minutes 04 seconds East a distance of 59.56 feet to a point for corner; THENCE North 02 degrees 48 minutes 56 seconds West a distance of 59.03 feet to a point for corner; THENCE North 02 degrees 50 minutes 22 seconds West a distance of 55.09 feet to a point for corner; THENCE North 08 degrees 54 minutes 24 seconds West a distance of 32.13 feet to a point for corner; THENCE North 26 degrees 56 minutes 24 seconds East a distance of 26.04 feet to a point for corner; THENCE North 11 degrees 31 minutes 56 seconds East a distance of 107.10 feet to a point for corner; THENCE North 14 degrees 15 minutes 12 seconds West a distance of 98.50 feet to a point for corner; THENCE North 36 degrees 47 minutes 33 seconds West a distance of 91.47 feet to the POINT OF BEGINNING, and containing 1.98 acres of land, more or less.

AND

A tract of land, situated in the Northwest Fractional Quarter of Section 30, Township 28 North, Range 21 West, Greene County, Missouri, and more particularly described as follows:

Commencing at the Southwest Corner of said Northwest Fractional Quarter of Section 30; THENCE South 89 degrees 49 minutes 28 seconds East a distance of 51.00 feet; THENCE North 00 degrees 27 minutes 32 seconds East a distance of 813.93 feet; THENCE South 89 degrees 32 minutes 28 seconds East a distance of 332.14 feet to the point of beginning; THENCE North 25 degrees 30 minutes 29 seconds East a distance of 110.00 feet to a point for corner; THENCE North 32 degrees 45 minutes 40 seconds East a distance of 55.44 feet to a point for corner; THENCE North 25 degrees 30 minutes 29 seconds East a distance of 55.00 feet to a point for corner; THENCE North 20 degrees 26 minutes 46 seconds East a distance of 79.34 feet to a point for corner; THENCE North 01 degrees 10 minutes 44 seconds West a distance of 79.14 feet to a point for corner; THENCE North 20 degrees 52 minutes 36 seconds West a distance of 79.40 feet to a point for corner; THENCE North 31 degrees 41 minutes 06 seconds West a distance of 296.40 feet to a point for corner; THENCE North 24 degrees 17 minutes 14 seconds East a distance of 95.49 feet to a point of curve; THENCE along a curve to the right having a radius of 305.00 feet a delta of 08 degrees 40 minutes 50 seconds, an arc length 46.21 feet and a chord which bears North 28 degrees 37 minutes 39 seconds East having a chord distance of 46.16 feet to a point on a line; THENCE South 22 degrees 34 minutes 15 seconds East a distance of 347.91 feet to a point for corner; THENCE South 56 degrees 19 minutes 32 seconds East a distance of 192.73 feet to a point for corner; THENCE South 23 degrees 51 minutes 46 seconds East a distance of 97.42 feet to a point for corner; THENCE South 62 degrees 11 minutes 24 seconds West a distance of 84.72 feet to a point for corner; THENCE South 12 degrees 07 minutes 41 seconds West a distance of 199.52 feet to a point for corner; THENCE South 52 degrees 19 minutes 18 seconds East a distance of 48.40 feet to a point for corner; THENCE South 37 degrees 40 minutes 42 seconds West a distance of 138.09 feet to a point on a curve; THENCE along a curve to the left having a radius of 205.00 feet a delta of 08 degrees 16 minutes 37 seconds, an arc length 29.61 feet and a chord which bears North 59 degrees 51 minutes 34 seconds West having a chord distance of 29.59 feet to a point of tangency; THENCE North 63 degrees 59 minutes 52 seconds West a distance of 166.19 feet to the POINT OF BEGINNING, and containing 2.71 acres of land, more or less.